Temporal constraints and device management for the Skill VRP: Mathematical model and lower bounding techniques *

Paola Cappanera¹, Cristina Requejo², and Maria Grazia Scutellà³

¹ Università degli Studi di Firenze, Italy. paola.cappanera@unifi.it ² Universidade de Aveiro, Portugal. crequejo@ua.pt ³ Università di Pisa, Italy. scut@di.unipi.it

July 31, 2020

Abstract

We study a generalization of the Skill VRP that incorporates time windows aspects, precedence and synchronization constraints. Specifically, we are given a logistic network where nodes correspond to customers, and where each customer requires a set of (partially ordered) operations. A set of technicians is available to perform such operations, and each technician is qualified to execute only a subset of them, depending on his skill. By referring to a specific context such as Health Care, customers are patients while technicians are caregivers. In a Field Service context, instead, customers are usually referred to as clients while technicians as field technicians. The innovative aspect is that some operations may require a special device, which must be transported at the customer site and must be present at the customer location together with a technician qualified to use it. Given technicians and for the special device, while respecting the skill compatibility between customers and technicians, and the time windows, precedence and synchronization constraints.

We propose a Mixed Integer Linear Programming (MILP) model for the generalized Skill VRP, and present some lower bounding techniques based on the proposed formulation. Preliminary computational experiments show that some lower bounding techniques may rapidly produce good lower bounds, thanks to quite effective valid inequalities. The returned percentage optimality gaps, estimated also thanks to a simple matheuristic, are in fact quite small for several scenarios of medium to large size, by encouraging the use of the proposed lower bounding techniques both as building blocks for designing exact approaches, and also as valuable tools to evaluate the efficacy of more sophisticated heuristic approaches to the problem.

Keywords: skill VRP; special device; precedence; synchronization; MILP model; lower bounding techniques.

^{*}This is a preprint of an article published in *Computers and Operations Research*. The final authenticated version is available online at: https://doi.org/10.1016/j.cor.2020.105054.

1 Introduction

Resource constrained routing and scheduling problems are very relevant in theory and practice, and thus they have attracted considerable attention by the scientific literature. In these problems, customers have specific requests that can be met by specialized resources, like technicians, vehicles and devices. In addition, the travel and the delivery of products or services to the customer locations have to be designed. The resources are usually classified as renewable (e.g. vehicles), non-renewable (e.g. spare parts), or doubly constrained (e.g. electric energy).

According to a recent review paper [27], the most popular problems in this field are the Skill VRP and the Technician Routing and Scheduling Problem. The Skill VRP has been introduced in [4] and further investigated in [5]. There, technicians with given skills must perform routes to serve customers, each requiring a set of skills to be served. The aim is to minimize the total routing costs while satisfying the constraints related to the required levels of skill. An extension to the Skill VRP has been studied in [32]. Another generalization has been proposed in [28], where the service of each customer requires one or more resources (e.g. technicians and equipment). There, the resources of each type are limited, and each of them must be assigned to a single route.

The Technician Routing and Scheduling Problem, introduced in [29], can be seen as another generalization of the Skill VRP dealing with multiple resources. Specifically, each technician has a set of skills and may in addition carry a set of tools and spare parts, while each customer requires a subset of them. Tools are renewable resources, whereas spare parts are non-renewable. The goal is to design minimum duration routes for the technicians so that each customer request is served by one technician with the required skills, tools and spare parts. Also in this case the resources of each type are limited, and each of them is carried out by a single technician. Variants to the Technician Routing and Scheduling Problem will be discussed in the section devoted to the literature overview. Interestingly, one of these problems has been the object of a recent challenge facilitated by VeRoLog, the EURO Working Group on Vehicle Routing and Logistics Optimization ([14]). Specifically, special tools for measuring milk quality have to be delivered to customers, at their request, over a multiple day time horizon. After the measurement, the tools have to be picked up again. The scheduling of these deliveries along the time horizon, and the routing for the planned deliveries and pickups, are the key decisions to address. The studied problem is indeed a simplification of a richer version faced by the service provider. In fact, as outlined in [23], the real problem includes also the scheduling and the routing of some inspectors, who should visit the customers while the measuring tools are present. Such inspectors are different than the technicians in charge of transporting the measuring tools, since they must possess specific qualifications. The need of technicians with different levels of skill, and the simultaneous presence of qualified technicians and devices, with the treatment of the underlying synchronization issues, would make the problem an even greater challenge to address.

The aim of this paper is to investigate in this direction of research, by addressing the management of technicians with different levels of qualification, and the simultaneous presence of a special device and of a properly qualified technician at some customer locations. Such aspects make the problem particularly significant in application contexts such as Home Care and Field Service. The relevance of considering special devices in Home Care is well described in [1],

where the authors present a case study to evaluate hospital readmission rates and mortality in elderly patients with chronic obstructive pulmonary disease. The treatment provided at home to such patients requires in fact to consider two types of materials: consumable materials, i.e. nonrenewable (e.g. oxygen replenishment), and specialized equipment, i.e. renewable (e.g. Doppler ultrasonograph and electrocardiograph). See also [31] for the description of a dimensioning problem in Home Care where the importance of managing devices is well outlined.

More in detail, with respect to the kernel Skill VRP problem presented in [4] and in [5], here multiple visits per customer are allowed, i.e. more than one technician may visit a customer in order to operate his required services, although each operation must be performed by a single technician. Furthermore, the generalization considers precedence constraints among the operations, time windows at the customers, and maximum workday duration constraints, which are not addressed in [4] and in [5]. Finally, the handling of a special device, which is required to perform some special operations, and which must be present at the customer location together with a technician qualified to use it, to the best of our knowledge constitutes an original contribution to the literature on Resource constrained routing and scheduling problems. The special device is moved by the technicians during their own tours, independently of their qualifications, and therefore non standard synchronization between the movement of the special device and the tours of the technicians must be imposed, giving rise to a peculiar tour for the special device, defined as a composition of some technician tour fragments. The resulting Skill VRP generalization appears thus to be relevant from an operational perspective, and it may constitute a basic block for interesting multiple day planning extensions.

The plan of the paper is the following. Section 2 proposes an overview of papers dealing with some of the characteristics of the generalized Skill VRP. Section 3 describes the studied problem, while Section 4 presents the Mixed Integer Linear Programming (MILP) model proposed for its mathematical formulation. Some simple, but computationally quite effective, valid inequalities to enhance the Linear Programming (LP) relaxation of the MILP model are presented in Section 5. Finally, Section 6 presents some lower bounding techniques, based on the proposed MILP formulation, and the results of computational experiments, showing that some lower bounding techniques, when enhanced via the proposed valid inequalities, may rapidly produce good lower bounds. The returned percentage optimality gaps, estimated also thanks to simple matheuristics, are in fact quite small for several scenarios of medium to large size, by encouraging the use of the proposed lower bounding techniques both as building blocks for designing exact approaches for the generalized Skill VRP, and also as valuable tools to evaluate the efficacy of more sophisticated heuristic approaches to the problem. Section 7 concludes the paper.

2 The literature

To the best of our knowledge, in the literature of Resource constrained routing and scheduling problems no paper has dealt with skill, precedence, synchronization and time windows aspects simultaneously with the management of a special device, which must be present at the customer location together with the technician in charge to use it. Rather, skill, precedence, synchronization and time windows constraints have been addressed, with different emphasis also depending on the considered application context. Also, delivery and pickup at the customer locations of renewable resources, such as tools and devices, and of non-renewable resources, such as spare parts, have been addressed, by disregarding however their synchronization with the visits of the technicians who are qualified to use them. Therefore, after listing recent reviews dealing with some of these issues in the Health Care application area, hereafter we shall provide a short overview of papers addressing some of these characteristics in the context of Resource constrained routing and scheduling problems.

Some Health Care reviews.

Cissé et al. [11] detail a comprehensive overview of recent models to Home Health Care routing and scheduling problems (HHCRSP). The authors analyze the existing literature according to the way the constraints and the objective function are formulated, provide an overview of methods developed to solve HHCRSP, and discuss future research directions.

Fikar and Hirsch [19] review relevant problem settings for HHCRSP. They consider singleperiod and multi-period Home Health Care problems and analyze objective functions, constraints and solution methods addressed in the literature.

Paraskevopoulos et al. [27] put in evidence that there exist several interesting variants of HHCRSP, the Skill VRP and the Technician Routing and Scheduling problems being the most prominent. The authors also show that maintenance activities and Home Health Care are the main areas where routing and scheduling of resources are crucial not only in terms of customer satisfaction, but also in terms of operational efficiency.

Time windows and preferences.

Cheng and Rich [10] consider the nurse rostering and routing problem. They propose a MILP formulation for the problem together with a heuristic approach. In particular, time windows are considered. A differentiation is made by considering full-time and part-time nurses, having different costs. Lunch breaks in different times and with different durations are addressed as well.

Berthels and Fahle [2] combine nurse rostering with vehicle routing aspects. They describe the software PARPAP developed for solving the considered problem. In particular, they use a Linear Programming model, constraint programming techniques and metaheuristics. The paper considers time intervals, preferences and other additional constraints as soft constraints.

Nickel et al. [26] consider scheduling and routing in Home Health Care over a weekly time horizon. Hard time windows are present. A two-stage solution approach is proposed which is based on a constraint programming heuristic and an adaptive large neighborhood search in order to improve the initial solution.

In a more general VRP setting, time windows are also addressed in [15]. There, a stochastic variant of the VRP with time windows is considered, where travel times are assumed to be stochastic, and some restrictions are placed on the probability of violation of time window constraints.

Precedence and synchronization.

Bredstrom and Ronnqvist [3] present a set partitioning model for the combined vehicle routing and scheduling problem with time windows and additional temporal constraints. The temporal constraints allow one to impose pairwise synchronization and pairwise temporal precedence between customer visits, independently of the vehicles. The authors propose an optimization based heuristic to solve real size instances.

Korsah et al. [24] address the problem of optimally assigning and scheduling a set of spatially distributed tasks to a fleet of vehicles working together to achieve a high-level goal, in domains where tasks may be related by precedence or synchronization constraints and might have a choice of locations at which they can be performed. Such problems may arise, for example, in disaster preparedness planning, transportation of people, and delivery of supplies. The authors present a novel mathematical model and describe how to solve it to optimality within a Branch-and-Price framework.

Rasmussen et al. [30] address scheduling and routing in Home Health Care, by considering time windows, soft preference constraints and temporal dependencies between the start times of the visits.

In the context of Home Health Care services, synchronization together with time window constraints are also treated in [16], [20] and [21]. In [16], a constructive heuristic is presented together with an iterated local search metaheuristic, whereas in [21] a general variable neighborhood search approach is provided. [20] extends the results in [21] by proposing variable neighborhood search approaches to coordinate multiple Health Care structures and multiple specialties handled by them.

In the same application context, we mention also [12] where some visits may require the presence of two staff members simultaneously, thus requiring visit synchronization. Hard and soft time windows are also taken into account together with multiple visits at some patient locations. A MILP model and a memetic algorithm featuring two specific crossover operators are proposed to solve the problem.

For a survey of vehicle routing problems with multiple synchronization constraints, related to spatial, temporal and load aspects, we refer to Drexl [13]. There, the authors present a classification of different types of synchronization, i.e. task synchronization, operation synchronization, movement synchronization, load synchronization and resource synchronization. Moreover, they comprehensively review the pertinent literature with respect to the application settings as well as with respect to successful solution approaches, both at an exact and at a heuristic level, by emphasizing that this kind of VRP problems are of practical relevance in many application areas, and they considerably complicate the use of standard VRP solution techniques.

Skills of staff.

Eveborn et al. [17] describes the LAPS CARE system, developed and used in Sweden for the daily planning of Home Care activities, by taking into account the skills of the staff members. The problem is formulated in terms of a set partitioning model, and solved via a repeated matching algorithm.

In a more general VRP setting, Cappanera, Gouveia and Scutellà [4] introduce the Skill VRP, where a set of tours, each one operated by a skilled technician, must be designed in such a way that each service requirement, asking for particular skills, is fulfilled by exactly

one technician [4, 5]. Various MILP models and related valid inequalities are proposed in [4]. Specifically, (i) the models are based on multicommodity flow variables and constraints, (ii) increasing levels of variable disaggregation allow one to strengthen the associated LP bounds, and (iii) the models strongly exploit skill information at the nodes of the logistics network, which is the key aspect differentiating the proposed models from more classical VRP models.

In [5], the authors formulate three MILP models, again characterized by an increasing level of variable disaggregation, and test them on a large suite of randomly generated instances. In particular, it is shown that the behavior of one of such models can be improved further by a cutting plane approach, so allowing to address large scale instances arising in challenging real application contexts.

Skill requirements and related constraints, in the context of assignment, routing and scheduling in Home Care, are also addressed in [6, 7, 9, 35]. See [8] for skill aspects in Home Care under uncertain patient demands.

Pickup and delivery of renewable and non-renewable resources.

In [29], each technician has a set of skills and may carry a set of tools and spare parts to the customers, each requiring a subset of these resources. Tools are renewable resources, whereas spare parts are non-renewable since they are consumed when the technician serves a request. The technicians start their tour from a depot with a set of spare parts and tools, and they also have the option to replenish their tools and spare parts at any time at the depot. The goal is to design minimum duration routes for the technicians so that each customer request is served by one technician, with the required skill, tools, and spare parts, within a given time window. The authors developed a parallel adaptive large neighborhood search algorithm, where a shared pool of promising solutions is maintained. A post-optimization procedure based on a set covering model is then used. Computational experiments on instances with up to 100 service requests are reported.

By considering a Home Health Care context, [25] studies the problem of delivering drugs and medical devices from the company's pharmacy to the patients, and special drugs from the hospital to the patients. Moreover, the pickup of biological samples and unused drugs and medical devices from the patients to the depot or to a medical lab is addressed. Each patient must be visited by one vehicle within a given time window, and vehicles are capacitated. Two MILP models are proposed together with a genetic algorithm and a tabu search method.

Always considering the Home Health Care setting, [33] addresses the case where a set of patient requests have to be accomplished by two types of caregivers in a one-day planning horizon. The first category of caregivers is responsible for primary tasks and to collect biological samples. These caregivers start their routes at a depot and end their working day at the initial depot right after the delivery of the collected samples at a laboratory. The perishability for transferring the biological samples to the laboratory is taken into account. Subsequently, the second category of caregivers performs synchronized services, that can be simultaneous or must respect precedence constraints, within a predetermined time window. In contrast to the first category, such caregivers do not visit the laboratory. The goal is to minimize the transportation cost and the idle time of the caregivers. Simulated annealing and tabu search algorithms are applied in two phases, i.e. to schedule primary tasks in the first phase, and to synchronize

services in the subsequent phase.

Finally, the already mentioned problem in [23] concerns a cattle company, that must regularly measure the milk quality at a number of farms, over a multiple day time horizon, by means of special measuring tools. Such tools have therefore to be delivered to the customers at their request. After a given number of days elapsed since delivery, the tools have to be picked up again. Tools are characterized by kind, size and availability. Each request involves only tools of the same kind. Vehicles transporting them are capacitated and the distance they can travel in one day is limited to a given maximum value. During the day, vehicles can visit the depot several times and load/unload tools. On each day, tools unloaded by one vehicle at the depot are not available to other vehicles. But, at the end of a day, the tools still on board of a vehicle are unloaded and made available to vehicles for the next day. Moreover, tools can be picked up at a customer and delivered at another customer during the same day.

It has to be decided on which day to serve each delivery request and, consequently, on which day to pick up the delivered tools. Furthermore, for each day in the planning horizon it has to be decided how to combine deliveries and pickups in the vehicle routes. The goal is to serve all the requests at a minimum cost. [23] provides a description of the solution methods proposed by the winners of the second and of the third place at the VeRoLog Solver Challenge 2016-2017. Specifically, the approach of the second ranked competitor involves sequences of low level heuristics, each of them having a transition probability to move to another low level heuristic. The third ranked method, instead, decomposes the problem into two independent subproblems, which are solved using a genetic algorithm, a deterministic routing algorithm, and a variable neighborhood descent algorithm to further improve the routes found by the deterministic routing algorithm.

Table 1 summarizes the different characteristics for the pickup and delivery of renewable and non-renewable resources literature.

Table 1: Home Health Care context (HHC), skilled technicians (ST), renewable resources (RR), non-renewable resources (NR), pickup resources (PR), replenish at depot (RD), timewindows (TW), capacitated vehicles (CV), precedence (P), synchronization (S).

Reference	year	HHC	ST	RR	NR	\mathbf{PR}	RD	TW	$_{\rm CV}$	Р	S
Pillac et al. [29]	2013		\checkmark	\checkmark	\checkmark		\checkmark	\checkmark			
Liu et al. [25]	2013	\checkmark		\checkmark	\checkmark	\checkmark		\checkmark	\checkmark		
Shahnejat-Bushehri et al. [33]	2019	\checkmark	\checkmark			\checkmark		\checkmark		\checkmark	\checkmark
Kheiri et al. [23]	2019			\checkmark		\checkmark	\checkmark	\checkmark	\checkmark		

3 The generalized Skill VRP

Let $G = (N_d, A)$ be a directed logistic network, with arc set A and node set $N_d = N \cup \{d\}$, where N is the set of the n customers and d is the depot node.

A set \mathcal{O}_j is associated with each node $j \in N$, indicating the set of the daily services or operations required by customer j. For example, in a Health Care context the required operations may include assistance with respect to personal needs such as getting out of bed, bathing, dressing, as well as medical services such as drug administration, blood test and ECG test. A partial ordering relation < is assumed to be associated with \mathcal{O}_j for each j, which specifies (partial) precedence relationships for the operations at j. Specifically, if o_1 and o_2 belong to \mathcal{O}_j for some j, $o_1 < o_2$ means that the operation o_1 must be accomplished before the operation o_2 is started in order to satisfy the requirements of node j. In a Health Care context, for example, dosing medicine must be performed before medication and blood test must be performed before a meal. A time window $[a_{jo}, b_{jo}]$ is associated with each customer and operation, indicating the time interval in which the operation o at customer j must start, $j \in N$ and $o \in \mathcal{O}_j$. Hereafter \mathcal{O} will denote the set of the available services or operations, i.e. $\mathcal{O} = \bigcup_{j \in N} \mathcal{O}_j$.

In addition to the normal services, just requiring the presence of a qualified technician, in our study we consider the situation in which a *special device*, say *s*, is needed for some of the operations. Let \mathcal{O}_s denote the subset of the operations needing such a special device. For example, to perform a blood test a centrifuge for blood stabilization is required, while a ECG machine must be available for the ECG test. An operation in \mathcal{O}_s can be performed by the assigned technician only if the special device is present when the operation is started, and the device must remain at the customer's home for the entire duration of the operation. W.l.o.g. we shall assume that each customer may require at most one operation in \mathcal{O}_s .

Let T be the set of the available technicians, and let S_t denote the *skill* or ability of technician t, expressed as the set of the operations that t is habilitated to perform. The skills constrain how the customer requirements can be operated: an operation o required by a customer j, i.e. $o \in \mathcal{O}_j$, can be operated only by a technician t possessing the corresponding skill, i.e. $o \in S_t$. Moreover, t can perform multiple operations at j. Concerning the transportation of the special device, we assume that its transportation is in charge of the technicians in T: different technicians may move s, during their own tours, to and from the location of the customers needing the special service; eventually, s can be transported via locations of customers do not requiring s, if this strategy can be useful for improving the overall service. Notice however that a technician visiting a customer has in any case to perform an operation at the customer site. By extending this strategy, the special device could be transported through sites other than the customer locations, like parkings and break centers, if present in the logistics network.

Given nonnegative technician dependent travelling costs c_{ij}^t , for each $(i, j) \in A$ and technician $t \in T$, we study the problem of defining a set of tours for the technicians, each one starting and ending at the depot d, in such a way that the set of the operations required by each customer j, i.e. \mathcal{O}_j , is fulfilled by qualified technicians (i.e. the skill constraints are satisfied), by respecting the time window constraints and the partial ordering relation associated with \mathcal{O}_j . This problem will be referred to as G-Skill VRP. The precedence constraints among the customer operations are imposed as follows: given the nonnegative travelling times τ_{ij} , for each $(i, j) \in A$, and given the operation duration δ_o^t , for each operation o and technician t, then if o_1 and o_2 belong to \mathcal{O}_j , and $o_1 < o_2$, the completion time of the technician having in care the operation o_2 at node j. Indeed, more general precedence constraints are considered in this paper, that include synchronization among operations as their special case. Such constraints will be better specified next. Constraints on the maximum workday duration of the technicians are considered as well. Precisely, the working time of technician t can not exceed his workday duration D^t , $t \in T$. The objective function, to be minimized, is the overall

travelling cost.

Next we summarize the introduced notation:

$N_d = N \cup \{d\}$	set of nodes, N is the set of the n customers and d is the depot node;
A	set of arcs;
O	set of the available services or operations;
\mathcal{O}_{j}	subset of the daily services or operations required by customer $j \in N$;
\mathcal{O}_s	subset of the operations needing a special device;
T	set of the available technicians;
S_t	subset of the operations that technician $t \in T$ is habilitated to perform;
$[a_{jo}, b_{jo}]$	time window associated with customer $j \in N$ and operation $o \in \mathcal{O}$;
c_{ij}^t	travelling cost for $(i, j) \in A$ and technician $t \in T$;
$ au_{ij}$	travelling time along $(i, j) \in A$;
δ_o^t	duration of operation $o \in \mathcal{O}$ when performed by technician $t \in T$;
D^t	workday duration of technician $t \in T$.

As outlined before, a specificity of the addressed problem, which makes it particularly challenging but also very complex, is the management of a special device, which is required for some of the operations. The special device is moved by the technicians during their own tours, and therefore non standard synchronization between the movement of the special device and the tours of the technicians must be imposed, giving rise to a peculiar tour for the special device, defined as a composition of some technician tour fragments. This is illustrated in Figures 1 and 2. The example refers to 5 customers and 4 operations, where $\mathcal{O}_1 = \{3\}$, $\mathcal{O}_2 = \{2,4\}$, $\mathcal{O}_3 = \{1,3,4\}$, $\mathcal{O}_4 = \{1\}$ and $\mathcal{O}_5 = \{1,2,3,4\}$. Each customer is represented by a circled node of the logistic network, and the required operations by smaller, internal, nodes, see Figure 1 (this modelling issue will be better explained in the next section). Operation 1 requires the special device. By assuming to have 3 technicians, who are qualified to perform the sets of operations $S_1 = \{1, 2, 4\}$, $S_2 = \{1, 2\}$ and $S_3 = \{1, 2, 3, 4\}$, respectively, a set of possible tours is depicted in Figure 2 (a).



Figure 1: Nodes of the network.

The three technicians have also in charge the movement of the special device, which is needed to perform operation 1. The route of the special device, resulting from the composition of the routes of the technicians moving it, is outlined in red in Figure 2 (b).



Figure 2: Routes of the technicians and of the special device.

4 The MILP model

Hereafter we shall present a MILP formulation to G-Skill VRP. Firstly we shall define the enlarged network used to better capture and model the peculiar problem characteristics. Then the various groups of constraints characterizing the MILP model, which is defined on the enlarged network, will be presented and commented, together with the objective function. The overall model will be reported in the appendix.

4.1 The enlarged network

The problem is formulated on an enlarged network $EG = (EV_d, EA)$, which is derived from the original logistic network $G = (N_d, A)$ as follows. For each customer $j \in N$ and each operation $o \in \mathcal{O}_j$ to be performed at node j, we introduce a customer/operation node $v = \langle j, o \rangle$ in the enlarged network. For each customer $j \in N$, we also introduce two fictitious nodes, $\langle j, s_{in} \rangle$ and $\langle j, s_{out} \rangle$. The two fictitious operations s_{in} and s_{out} associated with each customer j, which are added to \mathcal{O}_j , are used in order to establish the entrance to and the leaving from the customer location of the special device, as better specified next, and they are also included in the operation set S_t of each technician t, since each technician can transport the special device. The aim is to be able to capture, within the model, the entrance and then the leaving of the special device at a customer location, so allowing its synchronization with the movements of the technicians in charge of its transportation and, eventually, with the technician who needs to use it.

Based on these modelling issues, we define the set $EV = \{v = \langle j, o \rangle | j \in N, o \in \mathcal{O}_j\} \cup \{\langle j, s_{in} \rangle, \langle j, s_{out} \rangle | j \in N\}$, and the set $EV_d = EV \cup \{\langle d, o_d \rangle\}$. The latter includes the depot node d and a fictitious operation $o_d \in \mathcal{O}_d$ such that $o_d \in S_t$ for all technicians t. The fictitious operation o_d is introduced in order to habilitate each technician to visit d.

Concerning the set of arcs in the enlarged network, EA contains the arcs $\{(u, v), u = \langle i, o_1 \rangle, v = \langle j, o_2 \rangle | i, j \in N_d, o_1 \in \mathcal{O}_i \setminus \{s_{in}\}, o_2 \in \mathcal{O}_j \setminus \{s_{out}\}, (i, j) \in A, i \neq j\} \cup \{(u, v), u = \langle j, o_1 \rangle, v = \langle j, o_2 \rangle | j \in N, o_1, o_2 \in \mathcal{O}_j \setminus \{s_{in}, s_{out}\}, o_1 \neq o_2\} \cup \{(u, v) | u = \langle j, s_{in} \rangle, v = \langle j, o \rangle, j \in N, o \in \mathcal{O}_j \setminus \{s_{in}\}\} \cup \{(u, v) | u = \langle j, o \rangle, v = \langle j, s_{out} \rangle, j \in N, o \in \mathcal{O}_j \setminus \{s_{out}\}\}$. Notice that, according to the stated definitions, a complete subgraph is present in the enlarged network for each customer j, linking together all the operations required at j, except for the fictitious

operations s_{in} and s_{out} . Hereafter the arcs involving the nodes s_{in} and s_{out} will be also referred to as *special arcs*. The special arcs (u, v), with $v = \langle j, s_{in} \rangle$, are used only by the (unique) technician transporting the special device to customer j, if this movement is planned. In other words the special device, if required at j or moving along the network through node j, arrives at j exclusively via a special arc entering $\langle j, s_{in} \rangle$, and leaves j via a special arc outgoing $\langle j, s_{out} \rangle$. The technicians do not transporting the special device, instead, do not use the special arcs. In other words, fictitious nodes of type $\langle j, s_{in} \rangle$ and $\langle j, s_{out} \rangle$ are visited only by a technician transporting the special device.

The enlarged network is represented in Figure 3 (a). There, the little brown nodes represent the fictitious nodes $\langle j, s_{in} \rangle$ and $\langle j, s_{out} \rangle$ at each customer j. The tours of the technicians for the example depicted in Figure 2 (a) and the tour of the special device, depicted in Figure 2 (b), are represented, in terms of the enlarged network, in Figure 3 (b). The figure shows that the tour of the special device, depicted in red, is defined as the composition of the special arcs visited by the technicians, plus suitable movements inside the customer nodes.



Figure 3: Synchronization of the routes.

In the enlarged network, the travelling time associated with an arc $(u, v), u = \langle i, o_1 \rangle, v = \langle j, o_2 \rangle$, with $(i, j) \in A$ (and so $i \neq j$), is the travelling time along (i, j) in the original logistic network, i.e. τ_{ij} . Notice that this holds also for the special arcs. On the other hand, the travelling time associated with any arc (u, v) such that $u = \langle j, o_1 \rangle$ and $v = \langle j, o_2 \rangle$ for some customer j, i.e. with the arcs linking operation nodes related to a given customer, is zero. Moreover, each node $v = \langle j, o \rangle$ in the enlarged network inherits the service times associated with the operation o corresponding to the node, which depend on technician t, and were previously defined as δ_o^t , whereas the service time for the special node $\langle d, o_d \rangle$ and the fictitious nodes $\langle j, s_{in} \rangle$ and $\langle j, s_{out} \rangle$ is zero. Concerning the travelling costs in the enlarged network, which are dependent on the technicians, they are set as follows: $c_{uv}^t = c_{ij}^t$ for (u, v) such that $u = \langle i, o_1 \rangle, v = \langle j, o_2 \rangle$, and $(i, j) \in A$ (so, $i \neq j$). This holds true also for the special arcs. Otherwise, i.e. (u, v) is such that $u = \langle j, o_1 \rangle$ and $v = \langle j, o_2 \rangle$ for some customer j, then $c_{uv}^t = 0$ for all t.

Next we introduce the decision variables and the groups of constraints.

4.2 Groups of constraints and objective function

Regarding the formulation, recall that each customer $j \in N$ requires a set $\mathcal{O}_j \subseteq \mathcal{O}$ of operations to be performed, where $\mathcal{O} = \bigcup_{j \in N} \mathcal{O}_j$ will be used to denote the set of all the

required operations. Recall also that an operation $o \in \mathcal{O}_j$ can be performed only by a qualified technician, i.e. by a technician $t \in T$ such that $o \in S_t$, where $S_t \subseteq \mathcal{O}$ is the set of operations t can perform. Furthermore, at most one operation $o \in \mathcal{O}_j$ may require the special device, i.e. $o \in (\mathcal{O}_j \cap \mathcal{O}_s)$.

Always regarding the formulation, given an arc (u, v) in the enlarged network, with $u = \langle i, o_1 \rangle$ and $v = \langle j, o_2 \rangle$, define $T_{uv} = \{t \in T : o_1 \in \mathcal{O}_i \cap S_t, o_2 \in \mathcal{O}_j \cap S_t\}$ as the subset of the technicians who can perform both o_1 and o_2 , and so can move along (u, v). Also define EA_t as the subset of arcs where technician $t \in T$ can move along, that is the subset of arcs $(u, v) \in EA$, with $u = \langle i, o_1 \rangle$ and $v = \langle j, o_2 \rangle$, such that both o_1 and o_2 belong to S_t .

Assignment of operations to technicians. We start by introducing the group of binary variables modelling the assignment of the operations to the technicians. For each node $v = \langle j, o \rangle \in EV$ and $t \in T$ such that $o \in \mathcal{O}_j \cap S_t$, let:

$$z_v^t = \begin{cases} 1 & \text{if technician } t \text{ visits node } v = \langle j, o \rangle \text{ to perform the operation } o \text{ at customer } j \\ 0 & \text{otherwise.} \end{cases}$$

Any operation of a customer, except the fictitious ones, must be performed by exactly one technician; therefore the assignment variables must satisfy:

$$\sum_{t \in T \mid o \in S_t} z_v^t = 1 \qquad \qquad v = \langle j, o \rangle \in EV, o \neq s_{in}, s_{out}$$

$$\tag{4.1}$$

On the other hand, the special nodes (j, s_{in}) and (j, s_{out}) are visited by a technician only if the special device is required at j, or if j has just to be traversed to move the special device along the network. Therefore:

$$\sum_{t \in T} z_v^t \le 1 \qquad \qquad v = \langle j, s_{in} \rangle \in EV \text{ or } v = \langle j, s_{out} \rangle \in EV \qquad (4.2)$$

Routing of the technicians. Now we introduce the group of the binary variables modelling the routing of the technicians. For each $(u, v) \in EA$ and $t \in T_{uv}$:

$$x_{uv}^{t} = \begin{cases} 1 & \text{if } (u, v) \text{ belongs to the tour of technician } t \\ 0 & \text{otherwise.} \end{cases}$$

The routing variables must satisfy constraints (4.3) – (4.7). With a little abuse of the notation, hereafter the node $\langle d, o_d \rangle$ will be simply denoted as d.

$$\sum_{(u,v)\in EA_t} x_{uv}^t = z_v^t \qquad \qquad v = \langle j, o \rangle \in EV, \ t \in T, \ o \in S_t \qquad (4.3)$$

$$\sum_{(u,v)\in EA_t} x_{uv}^t = \sum_{(v,u)\in EA_t} x_{vu}^t \qquad v = \langle j, o \rangle \in EV, \ t \in T, \ o \in S_t$$
(4.4)

$$\sum_{(d,v)\in EA_t} x_{dv}^t = 1 \qquad t \in T \tag{4.5}$$

$$\sum_{(v,d)\in EA_t} x_{vd}^t = 1 \qquad t \in T \tag{4.6}$$

$$x_{uv}^t \in \{0, 1\}$$
 $(u, v) \in EA, \ t \in T_{uv}$ (4.7)

Constraints (4.3) guarantee that, if an operation o in \mathcal{O}_j is assigned to technician t, then t must visit node $v = \langle j, o \rangle$. Notice that this includes also the special nodes $\langle j, s_{in} \rangle$ and $\langle j, s_{out} \rangle$. Constraints (4.4) state that, if technician t enters a node v, then t must leave v. Notice that a technician may perform more operations at the same customer j, by moving among nodes of form (j, o), for different operations $o \in \mathcal{O}_j$, with travelling time zero, according to the definition of the enlarged network. Moreover, a technician may perform more operations at the same customer j by moving to different customers and then coming back to j. Therefore, multiple visits of a customer by the same technician are allowed.

Constraints (4.5) and (4.6) ensure that every technician leaving the depot to perform the assigned operations, then enters the depot after operating. Moreover, to model the case of technicians who are not assigned any operations, and therefore must remain at the depot, we assume that a fictitious arc (d, d), connecting the depot to itself, and having travelling time zero, belongs to the set EA_t for each t. In constraints (4.5)-(4.6), the routing variable x_{dd}^t will be set to 1 in case of inactivity of technician t, and to 0 otherwise. Constraints (4.7) define the variable domain.

Technician temporal constraints. In order to prevent subtours, that is tours which are disconnected from the depot, the proposed model uses temporal flow variables and standard flow conservation constraints, plus linking constraints to relate the flow variables to the routing variables. The temporal flow variables are used also to impose precedence and synchronization constraints among the customer operations, as better clarified next. Let us discuss the temporal flow variables and the related constraints in more detail.

Let y_{uv}^t , with $(u, v) \in EA$ and $t \in T_{uv}$, be a flow variable that specifies the cumulative time spent by technician t after traversing the arc (u, v), if (u, v) belongs to the tour of technician t. This cumulative time is defined as the sum of the travelling times associated with the arcs of the subpath from the depot to v, plus the durations of the operations performed by t at the nodes of this subpath, except the one operated at node v. Therefore, y_{uv}^t indicates the arrival time of t at v, if v is visited by t.

The temporal flow variables are defined in terms of the technician assignment variables z_v^t and of the technician routing variables x_{uv}^t :

$$\sum_{(d,v)\in EA_t} y_{dv}^t = \sum_{(d,v)\in EA_t} \tau_{dv} x_{dv}^t \qquad t\in T$$

$$\tag{4.8}$$

$$\sum_{(v,u)\in EA_t} y_{vu}^t = \sum_{(u,v)\in EA_t} y_{uv}^t + \delta_o^t z_v^t + \sum_{(v,u)\in EA_t} \tau_{vu} x_{vu}^t \qquad v = \langle j, o \rangle \in EV, t \in T$$
(4.9)

$$y_{uv}^t \le D^t x_{uv}^t \qquad \qquad t \in T, \ (u,v) \in EA_t \qquad (4.10)$$

$$y_{uv}^t \ge 0 \qquad \qquad t \in T, \ (u,v) \in EA_t \qquad (4.11)$$

Constraints (4.8) take into account that at the depot node d the duration of an operation is zero and therefore the cumulative time of the first visited customer, if any, accounts only for the duration of the travelling from the depot node to the customer node. The flow conservation constraints (4.9) define the total time accumulated by a technician t after the service at a customer node v as the cumulative time from the depot to v plus the duration of the operation at v plus the travelling time along the arc of the subpath leaving node v, if v belongs to the tour of t. Constraints (4.10) and (4.11) state that the cumulative time spent by technician t can be positive only on the arcs belonging to the tour of technician t. Constraints (4.10) play also another role into the model. In fact, they state that the total time of technician t in any point of his tour can not exceed D^t , i.e. the workday duration associated with t.

Notice that, in order to allow some waiting time along the technicians route, equalities (4.9) could be replaced by the inequalities

$$\sum_{(v,u)\in EA_t} y_{vu}^t \ge \sum_{(u,v)\in EA_t} y_{uv}^t + \delta_o^t z_v^t + \sum_{(v,u)\in EA_t} \tau_{vu} x_{vu}^t \qquad v = \langle j, o \rangle \in EV, \ t \in T.$$

Precedences and synchronization. By using the temporal flow variables, the precedence constraints among the customer operations can be modelled as follows:

$$\sum_{(u,v_1)\in EA_t} y_{uv_1}^t + \delta_{o_1}^t z_{v_1}^t \le \sum_{(u,v_2)\in EA_r} y_{uv_2}^r + D^t (1 - z_{v_2}^r)$$

$$v_1 = \langle j, o_1 \rangle, v_2 = \langle j, o_2 \rangle \in EV | o_1 < o_2, \ t, r \in T : o_1 \in S_t, o_2 \in S_r$$

$$(4.12)$$

Constraints (4.12) state that, for each customer j, if the operation o_1 must precede o_2 , and if technicians t and r have been selected, respectively, to operate o_1 and o_2 at node j, then the cumulative time of t at v_1 plus the time spent by t for operating o_1 must be less than or equal to the cumulative time of r at node v_2 . Notice that r = t is allowed, so modelling the case where the precedence relationship involves operations of j assigned to the same technician.

Indeed, the precedence constraints (4.12) can be generalized as follows. Let $g_{o_1o_2}^{j\min}$ be the minimum time that must elapse between the end of the operation o_1 and the beginning of the operation o_2 at customer $j \in N$, while $g_{o_1o_2}^{j\max}$ be the maximum allowed time between the end of the operation o_1 and the beginning of the operation o_2 , with $o_1, o_2 \in \mathcal{O}_j$. Then, we can impose that the gap between the completion time of operation o_1 at j and the beginning time of o_2 at j must lay between $g_{o_1o_2}^{j\min}$ and $g_{o_1o_2}^{j\max}$, for each customer j and any pair of operations $o_1, o_2 \in \mathcal{O}_j$:

$$g_{o_{1}o_{2}}^{j\min} - (D^{t} + g_{o_{1}o_{2}}^{j\min})(2 - z_{v_{2}}^{r} - z_{v_{1}}^{t}) \leq \sum_{(u,v_{2})\in EA_{r}} y_{uv_{2}}^{r} - (\sum_{(u,v_{1})\in EA_{t}} y_{uv_{1}}^{t} + \delta_{o_{1}}^{t} z_{v_{1}}^{t}) \leq g_{o_{1}o_{2}}^{j\max} + D^{r}(2 - z_{v_{1}}^{t} - z_{v_{2}}^{r})$$

$$v_{1} = \langle j, o_{1} \rangle, v_{2} = \langle j, o_{2} \rangle \in EV | t, r \in T : o_{1} \in S_{t}, o_{2} \in S_{r}$$

$$(4.13)$$

Notice that, depending on the setting of $g_{o_1o_2}^{j\min}$ and of $g_{o_1o_2}^{j\max}$, constraints (4.13) can be used to synchronize the operations o_1 and o_2 at node j. In case no synchronization is required, $g_{o_1o_2}^{j\min}$ can be set to a very negative value while $g_{o_1o_2}^{j\max}$ can be set to a very large value. **Time windows.** Concerning the time windows constraints, stating that the operation o at customer j must be started in the time interval $[a_{jo}, b_{jo}]$, for $j \in N$ and $o \in \mathcal{O}_j$ ($o \neq s_{in}, s_{out}$), they can be expressed as:

$$a_{jo}\left(\sum_{(u,v)\in EA_t} x_{uv}^t\right) \le \sum_{(u,v)\in EA_t} y_{uv}^t \le b_{jo}\left(\sum_{(u,v)\in EA_t} x_{uv}^t\right) \qquad v = \langle j,o\rangle \in EV(o \neq s_{in}, s_{out}), \ t \in T$$

$$(4.14)$$

Routing of the special device. Now, let us model the handling of the special device within the enlarged network. As previously outlined, the modelling idea is that the special device can arrive at customer j only through the node $\langle j, s_{in} \rangle$, thanks to a technician. Then, if the special device is required at j, it is transported directly to the requiring node $\langle j, o \rangle$, with $o \in \mathcal{O}_j \cap \mathcal{O}_s$, and, from there, after the completion of operation o, to the outgoing node $\langle j, s_{out} \rangle$, to leave customer j. Notice that the technician operating o at j may be different than the technician who have transported the special device at j. Also the technician who is in charge of moving the special device out of j may be different. Otherwise, i.e. the special device is not required at j, i.e. j is just used to move the device elsewhere, then the device moves directly from $\langle j, s_{in} \rangle$ to $\langle j, s_{out} \rangle$, to leave customer j. Also in this scenario the arrival and the leaving of the special device to/from j are in charge of a technician (possibly two different technicians).

In order to model the movement of the special device, let us denote by EA_s the set of the arcs along which the special device can move. This set contains: the arcs of form $(\langle i, s_{out} \rangle, \langle j, s_{in} \rangle)$ for each pair of different customers i and j such that an arc (i, j) exists in the original logistic network, modelling the movement of the special device from customer i to customer j; the arcs of form $(\langle j, s_{in} \rangle, \langle j, s_{out} \rangle)$, modelling the case where the special device is not required at j, but just traverses j; the arcs of form $(\langle j, s_{in} \rangle, \langle j, o \rangle)$ and $(\langle j, o \rangle, \langle j, s_{out} \rangle)$, modelling the case where the special device is required at j to perform the special operation o, with $o \in \mathcal{O}_j \cap \mathcal{O}_s$; and finally the arcs of form $(d, \langle j, s_{in} \rangle)$ and $(\langle j, s_{out} \rangle, d)$, modeling the leaving of the special device from the depot, and its coming back to the depot at the end of the working day. By extending the previous definition, all these arcs will be also referred to as special arcs. Then, let us introduce the following routing variables:

$$w_{uv}^{s} = \begin{cases} 1 & \text{if } (u,v) \text{ belongs to the tour of device } s, \text{ for } (u,v) \in EA_{s} \\ 0 & \text{otherwise.} \end{cases}$$

$$\sum_{(u,v)\in EA_s} w_{uv}^s = 1 \qquad \qquad v = \langle j, s_{in} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s \neq \emptyset$$
(4.15)

$$\sum_{(v,u)\in EA_s} w_{vu}^s = 1 \qquad \qquad v = \langle j, s_{out} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s \neq \emptyset \qquad (4.16)$$

$$w_{\langle j, s_{in} \rangle, \langle j, o \rangle}^{s} = w_{\langle j, o \rangle, \langle j, s_{out} \rangle}^{s} = 1 \qquad \qquad j \in N, o \in \mathcal{O}_{j} \cap \mathcal{O}_{s}$$

$$(4.17)$$

$$\sum_{(u,v)\in EA_s} w_{uv}^s \le 1 \qquad \qquad v = \langle j, s_{in} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s = \emptyset$$
(4.18)

$$\sum_{(v,u)\in EA_s} w_{vu}^s \le 1 \qquad \qquad v = \langle j, s_{out} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s = \emptyset$$
(4.19)

$$w^{s}_{(\langle j, s_{in} \rangle, \langle j, s_{out} \rangle)} \ge \sum_{(u, \langle j, s_{in} \rangle) \in EA_{s}} w^{s}_{(u, \langle j, s_{in} \rangle)} \qquad j \in N, \mathcal{O}_{j} \cap \mathcal{O}_{s} = \emptyset$$

$$(4.20)$$

$$\sum_{(d,u)\in EA_s} w_{du}^s = 1 \tag{4.21}$$

 $w^{s}_{\langle j, s_{in} \rangle, \langle j, o \rangle} = w^{s}_{\langle j, o \rangle, \langle j, s_{out} \rangle} \qquad \qquad j \in N, \ o \in \mathcal{O}_{j} \cap \mathcal{O}_{s}$

$$\sum_{(u,v)\in EA_s} w_{uv}^s = \sum_{(v,u)\in EA_s} w_{vu}^s \qquad \qquad v = \langle j, s_{in} \rangle \in EV \text{ or } v = \langle j, s_{out} \rangle \in EV \quad (4.23)$$

$$\sum_{\substack{(u,d)\in EA_s}} w_{ud}^s = 1$$

$$\sum_{t\in T} x_{uv}^t = w_{uv}^s$$

$$(u,v)\in EA_s, u = \langle i, s_{out} \rangle, v = \langle j, s_{in} \rangle, i \neq j$$

$$(4.24)$$

(4.22)

$$\sum_{t \in T} x_{(d,\langle j,s_{in}\rangle)}^{t} = w_{(d,\langle j,s_{in}\rangle)}^{s} \qquad (d,\langle j,s_{in}\rangle) \in EA_{s}$$

$$(4.26)$$

$$\sum_{t \in T} x_{(\langle j, s_{out} \rangle, d)}^t = w_{(\langle j, s_{out} \rangle, d)}^s \qquad (\langle j, s_{out} \rangle, d\rangle \in EA_s \tag{4.27}$$

$$w_{uv}^s \in \{0, 1\} \tag{4.28}$$

Constraints (4.15)-(4.24) are the routing constraints for the special device. In particular, constraints (4.15)-(4.17) guarantee the presence of the special device at each customer requiring it, whereas constraints (4.18)-(4.20) address the possible movement of the special device through the location of a customer does not requiring it. Constraints (4.21) and (4.24) consider the movement of the special device from and to the depot node, while (4.22) and (4.23) are the classical flow conservation constraints for the special device routing.

Constraints (4.25), (4.26) and (4.27), instead, ensure that the special device can move between different customers, and from/to a customer and the depot, if and only if a technician moves along the corresponding special arcs during his tour, by transporting the device. The tour of the special device, at the original logistic network level, is thus the composition of some technician tour fragments, related to those technicians having in charge the transportation of the special device. The enlarged network, instead, allows one a deeper view of the technicians and of the special device tours, by showing their behavior inside the customer nodes.

Precedence and synchronization for the special device. Here we introduce the temporal variables and constraints related to the special device. Hereafter D^s will denote the daily availability of the special device.

The temporal variables for the special device, y_{uv}^s , are defined only for the special arcs linking different customers in the tour of the special device, and for the special arcs of this tour linking customers to the depot. They are defined as a function of the temporal variables of the technicians according to the following observations, which are based on the previously stated assumption that the special device is moved along the network by means of technicians: (a) the arrival time of the special device at customer j coincides with the arrival time at node $\langle j, s_{in} \rangle$ of the (unique) technician visiting $\langle j, s_{in} \rangle$ (recall, in fact, that the special device enters customer jonly via the node $\langle j, s_{in} \rangle$); (b) the leaving time of the special device out of customer j coincides with the arrival time at node $\langle j, s_{out} \rangle$ of the (unique) technician visiting $\langle j, s_{out} \rangle$ (recall, in fact, that the special device leaves customer j only via the node $\langle j, s_{out} \rangle$). These modelling tools allow one to synchronize the arrival and the leaving time of the special device at customer j, and also to synchronize such times with the time of the technician in charge to use the special device at j, if j requires the special device. Formally:

$$\sum_{(u,\langle j,s_{in}\rangle)\in EA_s} y_{u\langle j,s_{in}\rangle}^s = \sum_{t\in T} \sum_{(u,\langle j,s_{in}\rangle)\in EA_s} y_{u\langle j,s_{in}\rangle}^t \qquad \qquad j\in N$$
(4.29)

$$\sum_{(u,\langle j,s_{in}\rangle)\in EA_s} y^s_{u\langle j,s_{in}\rangle} \le \sum_{(u,\langle j,o\rangle)\in EA_t} y^t_{u\langle j,o\rangle} + D^s(1-z^t_{\langle j,o\rangle})$$

$$i \in N \ t \in T \quad o \in \mathcal{O}_i \cap \mathcal{O}_o \cap S_t$$

$$(4.30)$$

$$\sum_{(u,\langle j,o\rangle)\in EA_t} y_{u\langle j,o\rangle}^t + \delta_o^t z_{\langle j,o\rangle}^t \le \sum_{r\in T} \sum_{(u,\langle j,s_{out}\rangle)\in EA_r} y_{u\langle j,s_{out}\rangle}^r$$

$$j\in N, \ t\in T, \ o\in\mathcal{O}_j\cap\mathcal{O}_s\cap S_t$$

$$(4.31)$$

$$\sum_{\langle j, s_{in} \rangle \rangle \in EA_s} y_{u\langle j, s_{in} \rangle}^s \leq \sum_{r \in T} \sum_{(u, \langle j, s_{out} \rangle) \in EA_r} y_{u\langle j, s_{out} \rangle}^r \qquad j \in N, \mathcal{O}_j \cap \mathcal{O}_s = \emptyset$$
(4.32)

$$y_{uv}^s \le D^s w_{uv}^s \tag{4.33}$$

$$y_{uv}^s \ge 0 \tag{4.34}$$

Constraints (4.29) state that the arrival time of the special device at customer j coincides with the arrival time of the (unique) technician transporting the device to j, i.e. visiting $\langle j, s_{in} \rangle$. Constraints (4.30) impose that the special device must arrive at customer j, requiring it, before the arrival of the technician who will use the special device at j, while (4.31) state that the use of the special device at j must be ended before the time when a technician will move the device out of j. Concerning the customers do not requiring the special device, i.e. those customer locations who may be used just to transport the device along the network, (4.32) state that the arrival time of the technician transporting the device at a customer must precede the leaving time of the one who will move out the device. Finally, constraints (4.33) guarantee that the total time of the special device tour can not exceed D^s and that the temporal variables for the special device may be positive only along the arcs used to transport it.

Objective function. The objective function, to be minimized, defines the overall routing cost:

$$\sum_{(u,v)\in EA} \sum_{t\in T_{uv}} c_{uv}^t x_{uv}^t \tag{4.35}$$

The overall model is reported in the appendix.

5 Valid inequalities

(u

In order to enhance the LP relaxation of the proposed formulation, the following valid inequalities, defined on the enlarged network, have been proposed. These inequalities, although quite simple, proved to be very effective from a computational perspective, as shown in the section devoted to the computational results. **Depot cut constraints.** These inequalities state that, if a visit is assigned to a technician t, then t must leave the depot node. Equivalently, the routing variable x_{dd}^t , modelling the inactivity of t through the fictitious link (d, d), must be set to 0.

$$x_{dd}^t \le 1 - z_v^t \qquad \qquad v = \langle j, o \rangle \in EV, \ t \in T, \ o \in S_t \tag{5.1}$$

See Figure 4 for a graphical explanation.



Figure 4: Depot cut constraints

2-3 cycle cut constraints. The 2 cycle cut constraints state that, if two operations of a certain customer j, other than the fictitious ones s_{in} and s_{out} , are assigned to a technician t, and thus the corresponding nodes in the enlarged network belong to his tour, then no cycle can exist in the tour of t comprising such nodes. Analogous inequalities can be stated by considering the case of three operations of customer j assigned to the same technician, leading to 3 cycle cut constraints.

$$x_{v_1v_2}^t + x_{v_2v_1}^t \le 1, \quad v_1 = \langle j, o_1 \rangle, \\ v_2 = \langle j, o_2 \rangle \in EV(o_1, o_2 \ne s_{in}, s_{out}), \quad t \in T, \quad o_1, o_2 \in S_t, \quad (5.2)$$

$$\begin{aligned} x_{v_1v_2}^t + x_{v_2v_1}^t + x_{v_3v_2}^t + x_{v_2v_3}^t + x_{v_1v_3}^t + x_{v_3v_1}^t &\leq 2 \\ v_1 &= \langle j, o_1 \rangle, v_2 = \langle j, o_2 \rangle, v_3 = \langle j, o_3 \rangle \in EV(o_1, o_2, o_3 \neq s_{in}, s_{out}), \ t \in T, \ o_1, o_2, o_3 \in S_t. \end{aligned}$$

$$(5.3)$$

These valid inequalities are illustrated in Figure 5.

2-3 aggregated cycle cut constraints. As only one technician operates each required operation, the 2-3 cycle cut constraints can be aggregated by technicians as follows:

$$\sum_{t \in T: o_1, o_2 \in S_t} (x_{v_1 v_2}^t + x_{v_2 v_1}^t) \le 1, \quad v_1 = \langle j, o_1 \rangle, \\ v_2 = \langle j, o_2 \rangle \in EV \ (o_1, o_2 \neq s_{in}, s_{out}), \tag{5.4}$$

$$\sum_{t \in T: o_1, o_2, o_3 \in S_t} (x_{v_1 v_2}^t + x_{v_2 v_1}^t + x_{v_3 v_2}^t + x_{v_2 v_3}^t + x_{v_1 v_3}^t + x_{v_3 v_1}^t) \le 2$$

$$v_1 = \langle j, o_1 \rangle, v_2 = \langle j, o_2 \rangle, v_3 = \langle j, o_3 \rangle \in EV \ (o_1, o_2, o_3 \neq s_{in}, s_{out}).$$

$$(5.5)$$



Figure 5: 2-3 cycle cut constraints

Aggregated inter-customers cycle cut constraints. They state that, given two customers i and j, if a technician t has to perform both an operation of i and an operation of j, then t can move either from i to j or from j to i. That is, cycles involving the corresponding nodes in the enlarged network are forbidden. Clearly, such valid inequalities can be generalized by considering more than two customers at a time.

$$\sum_{t \in T, o_1, o_2 \in S_t} (x_{v_1 v_2}^t + x_{v_2 v_1}^t) \le 1 \qquad v_1 = \langle j, o_1 \rangle, v_2 = \langle i, o_2 \rangle \in EV, i \neq j \qquad (5.6)$$

6 Computational experiments

In this section we report computational experiments performed to access the quality of the bounds obtained for G-Skill VRP. All the computational tests were performed using a processor Intel(R) Core(TM) i7-4750HQ CPU @ 2.00 GHz with 8GB of RAM and using the software Xpress 8.5 (Xpress Release 2018 with Xpress-Optimizer 33.01.05 and Xpress-Mosel 4.8.4) [18].

In Section 6.1 we provide details on the generation of the instances. In Section 6.2 we describe the procedures we applied to the models to obtain bounds to the problem. In Section 6.3 we report the computational results obtained for the generated instances, and discuss the main achievements that can be derived based on these results.

6.1 Instance description

We experimented two different sets of instances. The first set, comprising 60 instances of medium to large size, has been generated starting from the Euclidean-distance data sets tc that were proposed and used by Gouveia in [22] for the first time, and utilized since then in several computational testings. The name tc of the data sets refers to a central position of the depot in a grid. The second set, instead, is composed of 9 very large instances, and it has been generated starting from the TRSP instances used in Pillac et al.'s experimentation [29], which in turn are based on the Solomon benchmark [34]. We want to emphasize that the Technician Routing and

Scheduling Problem, investigated in [29], is different than G-Skill VRP. Therefore, Pillac et al.'s instances have been adapted to face with the specific characteristics of G-Skill VRP.

6.1.1 *tc*-based instances

We generated a total of 60 instances with |N| = 20 customers, |O| = 6 available operations and |T| = 3 technicians. Two of the technicians are skilled for 4 operations, while the third one is skilled for all the operations. Each technician is able to perform the special operation, i.e. operation 1, that requires the special device, and the working day duration is 480 minutes (8 hours) for all the technicians. Each customer requires 1 or 3 operations and the 20% of the customers require that their operations start in a given time window, either morning (first half of the working day duration) or afternoon (second half of the working day duration). Notice that, since in several application contexts the service delivered at the customer locations may be variable or uncertain, as remarked in [27], it may be reasonable to consider large time windows, such as morning and afternoon, rather than hard time windows, since these could be rarely satisfied in practice.

The 60 instances are divided into four sets of 15 instances each. Sets of instances have different percentages of the number of operations required per customer. Specifically, we have the following sets: (i) set 70-30 in which 70% of the customers (14 customers) require one operation and 30% of the customers (6 customers) require 3 operations, summing up to 32 operations, (ii) set 60-40 in which 60% of the customers require one operation and 40% of the customers require 3 operations (for a total of 36 operations), (iii) set 50-50 in which 50% of the customers require 3 operations (for a total of 40 operations), (iv) set 40-60 in which 40% of the customers require one operation and 60% of the customers require 3 operations (for a total of 44 operations).

We used five Euclidean data sets from [22], namely tc1, tc2, tc3, tc4 and tc5. For each one of the five Euclidean data sets, three different seeds (identified by the numbers 0, 1 and 2) were used to randomly assign operations to customers and technicians. The name of each generated instance reports the name of the original data set used (tc1, tc2, tc3, tc4 or tc5), the set it belongs to (70-30, 60-40, 50-50 or 40-60), and the random seed (0, 1 or 2). As an example, instance tc1-70-30-0 refers to the instance generated from the original instance tc1, belonging to the set 70-30, using the first seed (zero).

The service time δ_o^t of each one of the |O| = 6 available operations ranges from 10 to 30 minutes. The operation that requires the use of the special device, i.e. operation 1, has a service time of 10 minutes.

The travel time τ_{ij} between customer *i* and customer *j* is set to $0.3c_{ij}$ where c_{ij} is the Euclidean distance value from the data sets *tc* for nodes *i* and *j*. Also the traveling costs c_{ij}^t , which depend on the technicians, are defined using the values c_{ij} of the Euclidean data sets. Specifically, two technicians have the same travelling cost, which is equal to c_{ij} , for each pair of customers *i* and *j*, whereas the third technician, who is the most skilled among the technicians, has a travelling cost that is two times the cost of the other ones, i.e. $2c_{ij}$, for each pair of customers *i* and *j*.

As for the precedence constraints, we consider two cases: (i) one type of precedence con-

straints in which operation 2 must be performed before operation 3 at each customer requiring them, i.e. 2 < 3, and (ii) two types of precedence constraints in which operation 5 must always be performed before operation 6 in addition to the precedence constraint between operations 2 and 3, i.e. 2 < 3 and 5 < 6.

6.1.2 TRSP-based instances

We selected 9 TRSP instances from [29] which involve 100 customers and 25 technicians. In such instances, each technician has a skill and a set of tools and spare parts he can manage. On the other hand, each customer requires a skill and a set of tools and spare parts.

In order to adapt this different scenario to the G-Skill VRP context, we disregarded spare parts and interpreted tools as operations, i.e. each of the tools required by a customer in the original instance becomes an operation in our instance. Consequently, if a customer i requires a skill k and a tool o in the original TRSP instance, then i becomes a customer requiring skill k and operation o in the corresponding G-Skill VRP instance, and he can be served only by a technician having skill k and possessing o in his tool set. As an example, if i requires skill k and needs tools o_1 and o_2 in the original TRSP instance, in the corresponding G-Skill VRP instance i becomes a customer requiring two operations, i.e. o_1 and o_2 , both with skill k. The total number of operations in the G-Skill VRP instance is thus the total number of tool requirements in the original TRSP instance. Notice that the concept of skill has been generalized with respect to the one presented in Section 3, and the models have been adapted accordingly. In Section 3, skills correspond to operations, and a technician t can perform an operation o at a certain customer if and only if o belongs to set of operations t can perform. In the TRSP instances, instead, skills do not correspond to operations. A technician t can operate operation o requiring skill k at a customer if and only if t has skill k (according to the definition of skill in [29]), and its set of operations (i.e. his set of tools in the original TRSP instance) includes o.

One tool, i.e. tool 1, has then been selected to define the special operation, that is the operation which requires the special device. Recall that no special device is managed in [29].

The size of the TRSP-based instances is summarized in Table 2. The name of each instance is exactly the one as the original instance and it consists of three fields that identify respectively how requests at customers are generated, the length of the planning horizon (1 day in our case), and the instance number. In regards to the first field, requests can be randomly generated (R), organized in clusters (C), or generated combining both (RC). In addition to the instance name, columns in Table 2 report respectively the number of customers (|N|), the total number of operations required by overall customers, and the number of technicians (|T|). Notice that, since we disregarded spare parts, we removed all the customers requiring only spare parts. Similarly, technicians only managing spare parts have been disregarded. Despite of some customer and technician removals, the obtained TRSP-based instances are really very large, with a total number of operations ranging from 86 to 120.

Service times for the operations and workday durations for the technicians, which are not provided in the original instances, are defined as for the tc-based instances. In particular, the working day duration is 480 minutes (8 hours) for all the technicians and service times range in $\{10, 15, 20\}$. Technician costs depend on distance and skill. Specifically, we assume that for

Instance	N	$\sum_{j} \mathcal{O}_{j} $	T
C101	65	95	20
C102	77	120	22
C103	65	89	22
R101	56	87	21
R102	68	101	20
R103	63	95	21
RC101	59	86	23
RC102	64	98	21
RC103	68	103	17

Table 2: Size of TRSP-based instances.

each technician who is able to perform no more than three operations, the cost of arc (i, j) is exactly the distance between customer i and customer j, whereas for each technician who is able to perform more than three operations the cost of an arc is twice the distance.

Concerning time windows, which are not provided in the original instances, we still consider large time windows, i.e. a morning or an afternoon time window. Differently from what happens in the tc-based instances, where only the 20% of the customers has an associated time window, in the TRSP-instance we associated with each customer a morning or an afternoon time window, thus making the instances even more challenging. Finally, no precedence constraints are given.

6.2 Solving procedures

In this section, we describe how different variants of the MILP model (4.1)–(4.35) and several procedures have been used to provide lower and upper bounds to the G-Skill VRP. Specifically, we tested four models and three procedures as described in the following.

The first model, referred to as the *plain model* (*p* for short) is just the complete model whose feasible set is described by constraints (4.1)–(4.34). The second model, referred to as *model with cuts* (*c* for short) is the plain model enhanced with several valid inequalities. Namely, the depot cuts (5.1), the 2-3 aggregated cycle cuts (5.4)–(5.5) and the aggregated inter-customers cycle cuts (5.6). The third model, referred to as the *relax model* (*r* for short), is a relaxation of the plain model in which the special device is neglected. The relax model is thus defined by a subset of the constraints that characterize the plain model, namely constraints (4.1)–(4.14). Finally, the fourth model consists in the relax model equipped with the same valid inequalities used in model *c*, namely (5.1), (5.4)–(5.5), (5.6). This model is referred to as *relax model with cuts* (*x* for short). For each of these models, the solver is used to obtain the solution value of the linear programming relaxation and the values of the best lower and upper bounds returned by the Branch and Bound procedure within an *a priori* fixed maximum amount of computational time. Specifically, for each model *m* with *m* in {*p*, *c*, *r*, *x*}, *LP*_m represents the linear programming relaxation solution value, *LB*_m the best lower bound value and *UB*_m represents the best upper bound solution value. At this regard, notice that UB_r and UB_x are not necessarily upper bounds to G-Skill VRP. The name of the four alternative models and the constraints describing them are briefly recalled in Table 3.

Constraints	No Cuts	Cuts (5.1),(5.4),(5.5),(5.6)
(4.1)-(4.34)	р	с
(4.1)-(4.14)	r	х

Table 3: Model names and constraints describing them.

Preliminary computational results have shown that the relax model is more efficient than the plain model. A possible explanation of why this happens is that by eliminating the complicating constraints related to the special device every single subproblem in the Branch and Bound tree becomes easier to solve. As a result, more nodes of the enumeration tree are explored and this allows to find better lower bounds. Motivated by these preliminary tests, we defined two procedures aiming at improving the lower bound values. In the first procedure, referred to as p_1 , the lower bound value obtained by model x is used as a lower bound value to model c, thus implementing a warm start for model c. As before, LB_{p_1} and UB_{p_1} represent, respectively, the best lower and upper bound values returned by procedure p_1 .

In the second procedure, referred to as p_2 , as in the first procedure the lower bound value obtained by model x is used as a lower bound constraint, and in addition we dynamically add violated constraints from the sets (4.15)–(4.34) to the relax model. As before, LB_{p_2} and UB_{p_2} represent, respectively, the best lower and upper bound values returned by procedure p_2 . Notice that UB_{p_2} is not necessarily an upper bound to G-Skill VRP.

While the two procedures above described aim at improving the lower bound value, in some cases returning an upper bound as a collateral effect, the third procedure is a matheuristic, based on the LP solution of model c, that assigns all the nodes requiring the special operation o_s to one technician at a time, in an exhaustive way. Its aim is to rapidly estimate the quality of the lower bounding techniques proposed in this paper, thanks to its computational efficiency. Specifically, the LP solution of model c is used to rank the technicians according to the (possibly fractional) number of nodes of type $v = \langle j, o_s \rangle$ assigned to them in the LP solution, by considering a nonincreasing order. The first technician according to this ordering, say t_s , is selected, and all the nodes of type $v = \langle j, o_s \rangle$ are assigned to him. Therefore, the corresponding assignment variables $z_{(i,o_s)}^{t_s}$ are set to one within model c, and the plain model so simplified is used to generate a feasible solution with a time limit of 1800 seconds. Then, the next technician according to the considered ordering is selected, all the nodes of type $v = \langle j, o_s \rangle$ are assigned to him, and again the simplified model c is solved to get a feasible solution, by iterating until all the technicians have been considered, and halving the time limit for the simplified model c solution at each iteration. The resulting matheuristic, referred to as h for short, returns the best feasible solution found among the performed iterations. Its pseudocode is reported in Algorithm 1. By extending the previous notation, UB_h will denote the value of the solution returned by h, since it provides an upper bound to G-Skill VRP. Notice that the best lower bound value obtained by solving the simplified versions of model c along the heuristic iterations is not necessarily a lower bound to

G-Skill VRP.

Algorithm 1 The matheuristic for G-Skill VRP. 1: $TL := EmptyList(); \setminus$ list of technicians 2: SL := EmptyList(); \setminus list of feasible solutions 3: $\alpha := 1800; \setminus$ set time limit (seconds) 4: $\overline{z} = \arg \min LP_c$; \\ solve the LP of model c and get the solution 5: for all $t \in T$ do 6: if $S_t \cap \mathcal{O}_s \neq \emptyset \setminus$ technician t can perform the special operation then 7: TL.Insert(t); $\ell_t := \sum_{v = \langle j, o \rangle : j \in N, \ o \in \mathcal{O}_j \cap S_t \cap \mathcal{O}_s} \overline{z}_v^t$ 8: 9: end if 10: end for 11: TL.Sort(by nonincreasing ℓ_t); 12: while TL.IsNotEmpty() do $t_s := \text{TL.Pop}(); \setminus$ extract the first item from TL 13:for all $v = \langle j, o \rangle, \ j \in N, \ o \in \mathcal{O}_j \cap S_{t_s} \cap \mathcal{O}_s$ do 14:Set $z_v^{t_s} = 1$ in model c; 15:16:end for sol := Solve(c, α); \\ solve model c with time limit α 17:SL.Insert(sol); 18: $\alpha := \alpha/2;$ 19:20: end while 21: return(SL.BestItem()); \setminus return the best solution from SL

In the next section, detailed results relative to the four models, namely, p, c, r and x, and to the three procedures, namely p_1 , p_2 and h, are presented and discussed separately for the two sets of G-Skill VRP instances.

6.3 Computational results for *tc*-based instances

We start the section with a set of tables that compare pairs of models and/or procedures. Specifically, in Table 4 we compare models p and c, while in Table 5 we compare models r and x. These two tables allow to evaluate the impact of adding valid inequalities on the quality of the lower bounds obtained. Summary information on optimality gaps and computational times for models p, c, r and x is reported in Table 6. Then, after a discussion about the two procedures, p_1 and p_2 , which exploit the relax model with the aim of providing enhanced lower bounds, a lower bound comparison is proposed in Table 7 by reporting for each model, and separately for each set of instances, the average relative gap of the corresponding lower bound with respect to the best lower bound achieved. Information on the best lower and upper bounds obtained via the proposed models and procedures is finally summarized in Table 8, also considering the matheuristic approach h. Additional summary information is given also via a graphical representation. Computational results obtained when two types of precedence constraints are considered exhibit the same trend of the results obtained with one type of precedence constraint and are thus not reported.

Tables 4 and 5 report the same type of information. Specifically, the first column identifies the instance. Then, for each considered model or procedure, identified by a character in the subscript, columns LP display the value of the linear programming relaxation, columns LBdisplay the best lower bound value obtained within the time limit imposed, columns UB display the best upper bound value obtained within the time limit imposed and columns gap display the corresponding optimality gap between the lower and the upper bound found, i.e., $gap_m =$ $100 \times (UB_m - LB_m)/LB_m$ where the character m identifies the model or the procedure used. Additionally, in some tables, columns "time" report the computational time in seconds used to obtain the displayed values. A time limit of 10800 seconds was imposed.¹ This is the time used when the column "time" is omitted.

Comparison of the plain model with the model with cuts. Table 4 reports results that allow the comparison of the *plain model* with the *model with cuts*. The best results obtained within a time limit of three hours of computations are reported. As indicated before, for each of the two models, either p or c, the gap measures the relative distance between lower and upper bounds obtained with that specific model.

The direct comparison between columns LP_p and LP_c as well as between columns LB_p and LB_c highlights that the valid inequalities are largely effective in improving the lower bounds. Furthermore, valid inequalities allow to reduce the gap remarkably and they usually allow to increase the number of instances for which a feasible solution is found. Specifically, from Table 4 we get that, on the set 70-30, the average gap computed on all the instances for which a feasible solution is found decreases from 62.6% obtained with model p to 26.7% obtained with model c. In addition, the introduction of cuts allows to close the optimality gap for 4 instances over 15 and to determine a feasible solution for every instance. On the set 60-40, the average gap decreases from 119.7% (model p) to 25.8% (model c). In addition, model c allows to find a solution also on the 5 instances on which model p fails, thus determining a solution for all the instances in the set. The gaps decrease remarkably also on instances 50-50 (from 256.2% to 19.6%) and on instances 40-60 (from 255.6% to 66.2). However, on these two sets the introduction of cuts might on some instances prevent the solver from finding a feasible solution. This is probably due to the fact that on the most difficult instances in the test bed the introduction of all the cuts from scratch slows down the solver.

Comparison of the relax model with the relax model with cuts. Table 5 reports results that allow the comparison of the *relax model* with the *relax model with cuts* in terms of linear programming relaxation value, best lower bound value and best solution value obtained within a time limit of three hours of computation. Both the models neglect the constraints involving the special device and thus, as observed, their upper bound does not necessarily correspond to a feasible solution to the original problem.

¹In the Branch and Bound process, the solver stops only after a Branch and Bound node has been completely explored; therefore, sometimes the computational time can be bigger than the established time limit.

Noticeably, we observe that both the relax model with cuts and the relax model are able to obtain a solution for all the instances. The lower and upper bound values obtained with the relax model with cuts are significantly better than those obtained with the relax model. Further, Table 5 shows that the relax model with cuts is able to close the optimality gap for many instances (36 out of 60 for the one precedence constraint, and indeed the same 36 instances out of 60 for the two precedence constraints). Contrarily, the relax model is able to close the optimality gap only for three instances over 60 (all of them belonging to the 70-30 set). Thus, embedding the model with cuts, also in this case allows to speed up the solver. Overall, the introduction of cuts makes it possible to significantly reduce the optimality gap. On the 70-30, the average gap reduces from 17.2% when model r is used to 3.6% when model x is used. The same figures show that the gap decreases from 26.1% to 2.0% on the 60-40, from 51.3% to 3.1% on the 50-50, and from 111.7% to 5.8% on the 40-60.

When comparing the computational results of Table 5 with the corresponding results displayed in Table 4, we conclude that the presence of the special device makes the problem much harder to solve. In particular, the average computational time of model x halved in comparison with that required by model c.

Table 6 summarizes the average results for each of the four sets of instances, by reporting the average optimality gap and the average computational time obtained with models p, c, r and x. The average time of model p is not reported since the time limit was always reached.

Comparison of procedures p_1 and p_2 . We also compared the two procedures using the value of the relax model with cuts as a lower bound cut and then adding the special device constraints, either statically (in procedure p_1) or dynamically (in procedure p_2). The comparisons are done in terms of the best lower bound value and the best upper bound value obtained within a time limit of three hours of computation. Note that the upper bound values obtained with procedure p_2 are not necessarily upper bound values to G-Skill VRP problem since the maximum computational time can be reached before all the constraints have been added. As a case study, we considered only the instances for which the optimality gap has been closed by the relax model with cuts. As already said, this happens for 36 out of 60 instances.

We report that, noticeably, interesting results were achieved by procedure p_1 , which was able to certify the optimality of the bounds provided by model x for 7 out of 36 instances. In addition, it allowed to reduce the gap obtained by model x on three instances, namely, tc2-70-30-2, tc4-70-30-1, and tc2-50-50-2.

Comparison of the best lower and upper bounds. To better compare the lower bounds computed by means of models p, c, r and x, for each model, and separately for each set of instances, we determined the best lower bound achieved by all the suggested models and procedures, and calculated the average relative gap, over the 15 instances of each set, of the obtained lower bounds with respect to the best lower bound. This is reported in Table 7. Specifically, the quality of both the lower bound given by the linear programming relaxation, i.e. LP, and the best lower bound obtained by each model, i.e. LB, within the given maximum computational time are reported. Finally, Table 8 reports results that allow the comparison of the best values determined by the proposed models and procedures within the time limit of three hours of computation, also considering the upper bounds obtained by the matheuristic h. Precisely, column bestLB reports the best lower bound achieved by all the proposed models and procedures, while column bestUB is the value of the best feasible solution found, also considering the heuristic h. The specific model or procedure allowing to obtain the best lower bound (bestLBm) and the best upper bound (bestUBm) is also reported for each instance. To get more information on the behavior of the heuristic, columns UB_h report the values of the feasible solutions obtained by h, while columns time_h indicate the required computational times. The results displayed in Table 8 show that the proposed models and procedures allow to close the gap for 6 instances in the first two sets, and for one instance in the third set. Furthermore, the gaps are often quite small for the first three sets of instances, whereas they may be quite large for the last set, comprising the more difficult instances of this first test bed.

We conclude the discussion on the tc-based instances with a global overview of the results obtained. Specifically, Figures 6, 7, 8, and 9, respectively for the instances 70-30, 60-40, 50-50 and 40-60, compare the lower bounds obtained by the four alternative models, namely p, c, r, and x, and the best upper bound, i.e. bestUB, against the best lower bound obtained, i.e. bestLB. That is, the percentage gap between the four lower bounds and bestLB is reported, as well as the percentage gap between bestUB and bestLB. The figures support a series of main observations:

- the best lower bound is almost always given by the model which disregards the special device constraints, i.e. x, once enhanced with cuts; in several cases, this model is also able to close the optimality gap;
- the proposed valid inequalities, defined on the enlarged network, are quite effective in improving the quality of the lower bounds;
- the difficulty of obtaining good bounds seems to increase with the total number of operations considered, thus making the 40-60 instances the most challenging in this data set;
- on some instances to determine a good upper bound seems to be critical.

Some of the proposed lower bounding techniques appear thus to be a valuable tool to compute good lower bounds to G-Skill VRP. In several cases, in fact, especially when the total number of operations is not very large, the associated optimality gaps, evaluated also via a simple matheuristic approach, are quite small, showing the efficacy of the presented models and techniques.

6.4 Computational results for *TRSP*-based instances

We tested models p, c and x, as well as procedures p_1 and p_2 , to provide lower bounds to the *TRSP*-based instances. Due to the complexity and the very large size of these instances, no one of such models returned an upper bound as a collateral effect. Moreover, the matheuristic

	Table	T. Flai	in model	(p) vs m	ouer with	(cuts (c))		
Instances	LP_p	LB_p	UB_p	gap_p	LP_c	LB_c	UB_c	gap_c
tc1-70-30-0	367.8	438 /	830	80.8	491 5	487.9	780	60.5
tel 70-30-0	265.0	430.4 520.6	004 722	09.0	421.0	407.2	702	18.4
tal 70 20 2	201.9	401.2	501	20.2	411.2	561.0	561	10.4
tc1-70-30-2	271.4	491.2	10191	105.7	411.2	501.0	971	62.4
tc2-70-30-0	371.4	495.0	005	80.7	420.2	565.2	641	13.4
tc2-70-30-1	280.4	561.2	620	12.0	400.7	626.0	626	13.4
tc2-70-30-2	340.6	405.4	882	78.0	429.7	589.6	738	25.2
tc3 70 30 1	340.0	495.4	850	78.0	388.6	517 7	822	58.8
to2 70 20 2	252.0	405.0	664	22.7	417.1	552.2	624	12.0
tc4 70 30 0	411.5	543.0	004	00.7	417.1	681.1	1042	53.0
tc4-70-30-1	430.9	591.9	734	24.0	496.4	659.0	659	0.0
tc4-70-30-2	462.1	597.6	962	61.0	510.8	689.0	689	0.0
tc5-70-30-0	347 7	531.5	783	47.3	409.1	603.9	704	16.6
tc5-70-30-1	364.4	491.7	1056	114.8	405.1	601.4	854	42.0
tc5-70-30-2	351.1	480.0	878	82.9	407.2	560.0	762	36.1
100-10-00-2	001.1	400.0	0.0	02.0	401.2	000.0	102	00.1
tc1-60-40-0	359.3	504.8	848	68.0	405.9	566.0	785	38.7
tc1-60-40-1	360.4	494.8	-	-	421.2	573.6	823	43.5
tc1-60-40-2	348.9	535.1	-	-	398.4	613.9	829	35.0
tc2-60-40-0	396.1	522.1	-	-	451.8	599.5	705	17.6
tc2-60-40-1	374.9	480.2	1161	141.8	409.8	558.2	886	58.7
tc2-60-40-2	371.9	555.6	881	58.6	411.2	635.8	654	2.9
tc3-60-40-0	311.3	455.2	1082	137.7	370.3	544.3	795	46.1
tc3-60-40-1	301.9	448.6	1117	149.0	349.9	511.3	639	25.0
tc3-60-40-2	319.5	483.2	910	88.3	377.2	547.8	741	35.3
tc4-60-40-0	406.3	597.3	1834	207.0	475.4	706.9	762	7.8
tc4-60-40-1	461.4	597.1	1151	92.8	508.3	662.4	820	23.8
tc4-60-40-2	398.4	522.1	-	-	467.9	650.8	728	11.9
tc5-60-40-0	334.8	504.8	-	-	394.3	611.2	695	13.7
tc5-60-40-1	311.9	473.4	731	54.4	393.3	558.7	689	23.3
tc5-60-40-2	350.9	475.3	1424	199.6	405.7	623.1	651	4.5
tc1-50-50-0	347.8	517.8	_	_	399.6	578.2	_	_
tc1-50-50-1	355.3	481.0	-	-	408.7	561.4	-	-
tc1-50-50-2	375.8	503.7	_	_	423.2	629.6	734	16.6
tc2-50-50-0	387.2	521.1	_	_	433.4	608.7		10.0
tc2-50-50-1	391.0	518.0	_	_	430.4	591.6	_	_
tc2-50-50-1	320.3	469.9	1887	301.6	382.6	646.1	800	23.8
tc3-50-50-0	344.6	481.8	-	-	390.9	570.7	-	-
tc3-50-50-1	319.5	449.9	-	_	385.2	596.0	799	34.1
tc3-50-50-2	350.7	491.2	_	_	387.2	608.4		-
tc4-50-50-0	391.4	568.4	-	-	441.9	691.8	-	-
tc4-50-50-1	399.7	555.1	1607	189.5	455.3	660.5	758	14.8
tc4-50-50-2	389.8	526.5	1746	231.6	452.5	645.6	702	8.7
tc5-50-50-0	328.3	462.9	-		376.4	597.5	-	-
tc5-50-50-1	356.3	508.0	2042	302.0	408.4	620.2	-	-
tc5-50-50-2	337.1	470.9	-	-	419.2	621.4	-	-
- tc1-40-60-0	333.5	463.6	1573	239.3	385.3	625.2	1039	66.2
tc1-40-60-1	320.7	477.8	-	-	378.1	587.3	-	-
tc1-40-60-2	306.7	508.7	-	-	347.2	676.6	-	-
tc2-40-60-0	353.7	489.6	-	-	405.5	641.6	-	-
tc2-40-60-1	388.5	478.0	-	-	422.4	593.8	-	-
tc2-40-60-2	322.3	516.4	-	-	357.8	657.2	-	-
tc3-40-60-0	313.2	467.3	1715	267.0	364.7	616.2	-	-
tc3-40-60-1	307.6	451.6	-	-	384.1	571.8	-	-
tc3-40-60-2	335.6	442.1	-	-	370.7	608.3	-	-
tc4-40-60-0	343.6	480.2	-	-	396.5	694.3	-	-
tc4-40-60-1	370.5	517.6	1866	260.5	440.4	660.6	-	-
tc4-40-60-2	386.6	513.7	-	-	432.7	671.6	-	-
tc5-40-60-0	313.1	464.1	-	-	401.9	617.0	-	-
tc5-40-60-1	343.7	462.8	-	-	410.9	617.8	-	-
tc5-40-60-2	328.4	484.4	-	-	366.2	607.3	-	-

Table 4: Plain model (p) vs model with cuts (c).

Table 5: Relax model (r) vs relax model with cuts (x)).
---	----

	-		J. Rela	model	(7) vs rela	ix model	with cuts	(x).		
Instances	LP_r	LB_r	UB_r	gap_r	time_r	LP_x	LB_x	UB_x	gap_x	$time_x$
tc1-70-30-0	357.0	498.9	596	19.5	10805	415.7	562.0	562	0.0	8026
tc1-70-30-1	303.1	516.7	659	27.5	10802	338.4	637.0	637	0.0	254
tc1-70-30-2	307.9	533.1	558	4.7	10802	380.1	558.0	558	0.0	34
tc2-70-30-0	335.1	520.2	729	40.1	10804	390.0	546.0	673	23.3	10817
tc2-70-30-1	335.8	541.1	572	5.7	10805	386.9	572.0	572	0.0	92
tc2-70-30-2	297.7	624.0	624	0.0	8267	344.4	624.0	624	0.0	26
tc3-70-30-0	299.4	510.2	730	43.1	10802	345.4	623.9	676	8.4	10871
tc3-70-30-1	288.2	505.5	662	31.0	10802	367.5	572.3	636	11.1	11032
tc3-70-30-2	336.5	503.9	603	19.7	10808	394.6	595.0	595	0.0	37
tc4-70-30-0	369.7	648.7	814	25.5	10801	428.6	719.9	797	10.7	10808
tc4-70-30-1	369.5	654.0	654	0.0	2185	482.3	654.0	654	0.0	20
tc4-70-30-2	450.0	633.0	691	9.2	11899	487.7	689.0	689	0.0	75
tc5-70-30-0	309.0	612.0	612	0.0	5289	362.4	612.0	612	0.0	53
tc5-70-30-1	333.1	588.9	682	15.8	10802	399.1	676.0	676	0.0	4867
tc5-70-30-2	314.8	535.4	623	16.4	10803	390.7	619.0	619	0.0	521
tc1-60-40-0	309.0	549.9	648	17.8	10807	374.0	598.4	638	6.6	10809
tc1-60-40-1	305.7	522.9	655	25.3	10806	361.4	583.0	583	0.0	114
tc1-60-40-2	308.5	556.6	764	37.3	10814	362.3	625.2	655	4.8	11334
tc2-60-40-0	337.6	539.9	698	29.3	10802	401.9	642.0	642	0.0	889
tc2-60-40-1	335.0	517.8	646	24.8	10808	387.2	599.0	599	0.0	447
tc2-60-40-2	304.2	616.5	654	6.1	10809	351.3	654.0	654	0.0	156
tc3-60-40-0	288.3	548.9	728	32.6	10802	344.3	671.0	671	0.0	2936
tc3-60-40-1	263.8	445.3	603	35.4	10802	326.4	540.0	588	8.9	10811
tc3-60-40-2	292.6	514.6	706	37.2	10812	361.7	592.6	651	9.9	11161
tc4-60-40-0	386.4	661.3	749	13.3	10807	445.4	743.0	743	0.0	122
tc4-60-40-1	398.3	623.8	786	26.0	10843	490.0	706.0	706	0.0	2535
tc4-60-40-2	350.4	574.5	735	27.9	10807	436.1	681.0	681	0.0	192
tc5-60-40-0	289.0	591.9	658	11.2	10802	361.2	651.0	651	0.0	167
tc5-60-40-1	266.0	527.9	630	19.3	10806	376.6	605.0	605	0.0	495
tc5-60-40-2	307.3	494.7	730	47.6	10803	366.3	644.0	644	0.0	182
tc1-50-50-0	302.7	523.0	849	62.3	10807	351.2	608.8	625	27	10804
tc1-50-50-1	269.6	496.0	760	53.2	10844	346.4	606.0	606	0.0	2988
tc1-50-50-2	269.0	543.1	753	38.6	10802	333.7	643.0	643	0.0	2300
tc2-50-50-0	345.3	546.0	684	25.3	10809	398.9	645.0	645	0.0	1587
tc2-50-50-1	339.4	525.3	749	42.6	10817	379.9	645.0	645	0.0	1302
tc2-50-50-2	295.5	597.3	718	20.2	10803	353.1	668.0	668	0.0	127
tc3-50-50-0	298.8	484.3	996	105.7	10806	353.3	629.4	662	5.2	10803
tc3-50-50-1	286.2	484.1	720	48.7	10811	356.8	638.9	664	3.9	10812
tc3-50-50-2	317.9	538.8	818	51.8	10806	358.9	638.5	732	14.6	10823
tc4-50-50-0	331.0	600.5	884	47.2	10818	398.3	761.0	761	0.0	960
tc4-50-50-1	327.9	538.9	852	58.1	10806	412.5	683.4	703	2.9	10806
tc4-50-50-2	330.0	559.9	744	32.9	10802	434.7	653.0	653	0.0	119
tc5-50-50-0	293.9	551.6	877	59.0	11447	354.6	649.4	708	9.0	10809
tc5-50-50-1	326.9	521.8	816	56.4	10944	381.2	682.0	682	0.0	4084
tc5-50-50-2	313.8	495.1	831	67.8	10801	389.1	683.4	740	8.3	10804
tal 40.60.0	282 4	522.0	860	61 7	10802	222 1	652 4	750	15.0	10052
tc1-40-60-0	202.4	492 0	700	60.0	10802	330.0	598.3	660	11.0	10900
tc1 40 60 2	200.0	495.9	1058	113.3	10844	200.9	721.3	760	6.6	10801
tc2-40-60-0	323.6	511.3	1184	131.6	10804	372.0	679.0	679	0.0	479
tc2-40-60-1	324.3	508.6	775	52.4	10802	363.8	650.0	650	0.0	3843
tc2-40-60-2	257.6	558.2	1113	99.4	10803	307.4	734.7	802	9.2	10803
tc3-40-60-0	258.5	460.8	971	110.7	10809	308.5	691.0	691	0.0	5920
tc3-40-60-1	277.5	481.4	791	64.3	10811	354.8	605.8	662	9.3	10801
tc3-40-60-2	281.7	466.0	1266	171.7	10824	324.8	679.8	691	1.6	10803
tc4-40-60-0	289.4	526.9	947	79.7	10802	362.1	709.0	709	0.0	38
tc4-40-60-1	306.0	525.2	938	78.6	10807	374.1	698.0	698	0.0	1654
tc4-40-60-2	319.2	541.1	1171	116.4	10811	395.3	746.4	760	1.8	10802
tc5-40-60-0	258.6	461.0	1344	191.5	10815	349.9	725.0	725	0.0	1824
tc5-40-60-1	277.9	485.4	1457	200.2	10810	348.8	653.2	780	19.4	10812
tc5-40-60-2	265.1	475.3	1161	144.3	10801	338.8	685.6	766	11.7	10815

Table 6: Comparing average gaps and computational times.

	Table 0. Comparing average gaps and computational times.										
Instances	$\operatorname{avg}(gap_p)$	$\operatorname{avg}(gap_c)$	$\operatorname{avg}(\operatorname{time}_{c})$	$\operatorname{avg}(gap_r)$	$\operatorname{avg}(\operatorname{time}_r)$	$\operatorname{avg}(gap_x)$	$\operatorname{avg}(\operatorname{time}_x)$				
70-30	62.6	26.7	8288	17.2	9765	3.6	3835				
60-40	119.7	25.8	10800	26.1	10800	2.0	3490				
50-50	256.2	19.6	10800	51.3	10800	3.1	5794				
40-60	255.6	66.2	10800	111.7	10800	5.8	7410				

Table 7: Comparing lower bounds with respect to the best lower bound.

	Table	1. Compar	ing lower b	ounds with	respect to t	ne best lowe	r bound.	
Instances	gap_{LP_p}	gap_{LB_p}	gap_{LP_c}	gap_{LB_c}	$gap_{LP_{T}}$	gap_{LB_r}	gap_{LP_x}	gap_{LB_x}
70-30	39.07	16.72	31.44	4.77	45.87	9.06	36.02	0.11
60-40	43.26	19.67	34.51	5.95	50.29	13.12	39.72	0.00
50-50	45.08	23.41	36.94	6.19	52.71	18.56	43.01	0.05
40-60	50.30	29.28	42.45	7.14	58.77	26.24	49.32	0.00

Table 8: Best lower and upper bounds and related optimality gaps, also considering matheuristic h.

Instances	UB_h	time_h	bestLB	bestLBm	bestUB	bestUBm	gap
tc1-70-30-0	567	3848	562.0	x	567	h	0.9
tc1-70-30-1	648	177	637.0	x	648	h	1.7
tc1-70-30-2	586	36	561.0	с	561	с	0.0
tc2-70-30-0	712	5404	546.0	x	712	h	30.4
tc2-70-30-1	586	80	572.0	x	586	h	2.4
tc2-70-30-2	629	26	626.0	c	626	c	0.0
tc3-70-30-0	706	5405	623.9	x	706	h	13.2
tc3-70-30-1	622	1479	572.3	<i>x</i>	622	h	8.7
tc3-70-30-2	603	57	595.0	<i>x</i>	603	h	1.3
tc4-70-30-0	770	4456	719.9		770	h	7.0
tc4-70-30-1	678	37	659.0	2	659	C C	0.0
tc4 70 30 2	680	26	689.0	<i>c m</i>	680	c h	0.0
tc4-70-30-2	676	20	612.0	<i>c</i> , <i>x</i>	676	<i>c</i> , <i>n</i>	10.5
tc5-70-30-0	670	1100	676.0	r, x	670	<i>n</i>	10.5
tc5-70-30-1	679	1192	676.0	x	679	n	0.4
tco-70-30-2	628	406	619.0	x	628	n	1.5
1 20 40 0	0.40	0100	500 4			,	0.0
tc1-60-40-0	648	3160	598.4	x	648	h	8.3
tc1-60-40-1	633	446	583.0	x	633	h	8.6
tc1-60-40-2	651	1186	625.2	x	651	h	4.1
tc2-60-40-0	661	230	642.0	x	661	h	3.0
tc2-60-40-1	608	298	599.0	x	608	h	1.5
tc2-60-40-2	654	116	654.0	x	654	c, h	0.0
tc3-60-40-0	689	1471	671.0	x	689	h	2.7
tc3-60-40-1	583	455	540.0	x	583	h	8.0
tc3-60-40-2	654	5402	592.6	x	654	h	10.4
tc4-60-40-0	746	107	743.0	x	746	h	0.4
tc4-60-40-1	727	411	706.0	x	727	h	3.0
tc4-60-40-2	687	406	681.0	x	687	h	0.9
tc5-60-40-0	657	289	651.0	x	657	h	0.9
tc5-60-40-1	612	781	605.0	x	612	h	1.2
tc5-60-40-2	644	162	644.0	x	644	h	0.0
tc1-50-50-0	686	5404	608.8	x	686	h	12.7
tc1-50-50-1	673	3630	606.0	x	673	h	11.1
tc1-50-50-2	672	235	643.0	- T	672	h	4.5
tc2-50-50-0	659	531	645.0	x x	659	h	2.0
tc2 50 50 1	603	3018	645.0	<i>x</i> <i>x</i>	603	h	7.4
te2 50 50 2	672	05	668.0	<i>x</i>	672	h	0.7
tc2-50-50-2	013	90 E 40.4	620.4	<i>x</i>	075	<i>n</i>	0.7
103-50-50-0	670	0404 4260	629.4	<i>x</i>	670	<i>n</i>	21.9
tc3-50-50-1	670	4360	638.9	x	670	n	4.9
tc3-50-50-2	991	6301	638.5	x	991	h	55.2
tc4-50-50-0	970	5405	761.0	x	970	h ,	27.5
tc4-50-50-1	716	3299	683.4	x	716	h	4.8
tc4-50-50-2	711	3955	653.0	x	702	c	7.5
tc5-50-50-0	829	5451	649.4	x	829	h	27.7
tc5-50-50-1	696	1400	682.0	x	696	h	2.1
tc5-50-50-2	748	5403	683.4	x	748	h	9.5
tc1-40-60-0	996	6301	652.4	x	996	h	52.7
tc1-40-60-1	713	5403	598.3	x	713	h	19.2
tc1-40-60-2	957	6308	721.3	x	957	h	32.7
tc2-40-60-0	681	4594	679.0	x	681	h	0.3
tc2-40-60-1	688	5076	650.0	x	688	h	5.8
tc2-40-60-2	1326	6302	734.7	x	1326	h	80.5
tc3-40-60-0	807	5441	691.0	x	807	h	16.8
tc3-40-60-1	738	5405	605.8	x	738	h	21.8
tc3-40-60-2	738	5404	679.8	<i>x</i>	738	h.	8.6
tc4-40-60-0	710	2088	709.0	<i>x</i>	710	h	0.1
tc4-40-60-1	710	710	698.0	T.	710	 Ь	17
tc4 40 60 9	970	5/10	746 4	л Т	870	n k	16.6
tc5 40 60 0	1079	691E	795.0	x	1079	n k	47.0
tco-40-00-0	1072	6300	120.0	x	1072	11	41.9
tc5-40-60-1	946	6302	053.2	x	946	h	44.8
tc5-40-60-2	780	5407	685.6	x	780	h	13.8



Figure 6: 70-30 instances: LB of each model and bestUB are compared against bestLB.



Figure 7: 60-40 instances: LB of each model and bestUB are compared against bestLB.

h failed in computing feasible solutions. We thus generalized h, in an attempt to determine upper bounds also for this second set of instances.

The idea is to handle the huge size of the TRSP-based instances by decomposition, i.e. partitioning the set of the customers of each instance into three subsets: i) a subset named *special*, which includes all of the customers requiring the special device, thus extending the philosophy of matheuristic h, where all the special operations are dealt together; ii) a subset named *morning*, which is composed of customers with a morning time window; and iii) a subset called *afternoon*, comprising customers with an afternoon time window. A generalized bin packing problem is solved for each subset of customers, in order to determine a subset of technicians able to serve them. Three smaller and independent G-Skill VRP subproblems are thus derived, for the special, the morning and the afternoon customer subsets, respectively. The three subproblems are then solved in cascade by model c, along the lines of matheuristic h, by setting a time limit of 2000 seconds for each subproblem. A solution to the overall TRSP-based instance is then obtained by combining the solutions found for the three subproblems.



Figure 8: 50-50 instances: LB of each model and bestUB are compared against bestLB.



Figure 9: 40-60 instances: LB of each model and bestUB are compared against bestLB.

Figure 10 shows, for each instance, the quality (relative gap) of the lower bounds provided by models p and c with respect to the best lower bound that for this set of instances is always given by procedure p_1 . On this set of instances, procedure p_2 is in fact dominated by procedure p_1 and thus the relative results are not reported. Similarly, we do not report the results of model x, since they are enhanced by p_1 . Figure 10, for each model, reports both the quality of the lower bound given by the linear programming relaxation (LP) and the quality of the best lower bound returned when the time limit is reached (LB). The time limit has been fixed to 3 hours; further computational results obtained with a time limit of six hours demonstrated that there is no substantial improvement when the time limit is extended. The results shown allow to make some interesting observations. First, as for the tc-based instances, the introduction of cuts (model c vs model p) drastically increases the quality of the lower bounds. Specifically, comparing LB_p and LB_c , we observe that the introduction of valid inequalities makes it possible to reduce the gap of 27.3% on average. Besides, their introduction seems to be even more crucial than for the tc-based instances. In fact, if we consider model p, we observe that there are instances for which the lower bound obtained after three hours of computation is the same as the initial lower bound given by linear programming (see instances C102 and R103 when model p is used), and in general, the improvement between LP_p and LB_p is still limited. Second, the difficulty in solving the instances does not seem to depend on how the requests are generated (R, C or RC), i.e. quite similar trends are observed on each instance of this set. Unfortunately, the quality of the solutions provided by the matheuristic we proposed is not acceptable. The matheuristic fails in one case over 9 instances and for the remaining ones the average relative gap between the upper bound and the best lower bound is about 164 %. Only a few subproblems (special, morning, afternoon) are solved to optimality and the average computational time on the 8 instances is 4316.35 seconds. This result comes at no surprise since several attempts made to extend the matheuristic used for the *tc*-based instances revealed that the *TRSP*-instances are really challenging to solve. Thus, the results obtained stimulate future research in the direction of improving the upper bounds.



Figure 10: Comparing LP and LB of models p and c to bestLB provided by procedure p_1 .

7 Conclusions

We have studied a generalization of the Skill VRP problem, named G-Skill VRP, where multiple visits per customer are allowed in addition to consider precedence constraints among the operations, time windows at the customers, and maximum workday duration constraints. Furthermore, the handling of a special device, which is required to perform some special operations, and which must be present at the customer location together with the technician in charge to use it, constitutes an original and relevant contribution to the literature on VRPs. The resulting Skill VRP generalization appears thus to be very relevant from an operational perspective, and it may constitute a basic block for interesting multi-day planning extensions.

We have proposed and empirically investigated some models and procedures to G-Skill VRP, with the aim of determining good bounds to the problem in an efficient way. The results of a wide computational experimentation show that some of the proposed techniques appear to be a valuable tool to compute good lower bounds to G-Skill VRP. In several cases, in fact, especially when the total number of operations is not so large, the associated optimality gaps, evaluated also via a simple matheuristic approach, are indeed rather small.

We plan to extend the study of G-Skill VRP by proposing more sophisticated heuristic approaches to the problem, to be evaluated also thanks to the presented models and techniques.

8 Appendix: the overall MILP model

Objective function.

$$\min \quad \sum_{(u,v)\in EA} \sum_{t\in T_{uv}} c_{uv}^t x_{uv}^t \tag{8.1}$$

Assignment of operations to technicians.

$$\sum_{T \mid o \in S_t} z_v^t = 1 \qquad \qquad v = \langle j, o \rangle \in EV, o \neq s_{in}, s_{out}$$
(8.2)

$$\sum_{t \in T} z_v^t \le 1 \qquad \qquad v = \langle j, s_{in} \rangle \in EV \text{ or } v = \langle j, s_{out} \rangle \in EV \qquad (8.3)$$

Routing of the technicians.

$$\sum_{(u,v)\in EA_t} x_{uv}^t = z_v^t \qquad \qquad v = \langle j, o \rangle \in EV, \ t \in T, \ o \in S_t \qquad (8.4)$$

$$\sum_{(u,v)\in EA_t} x_{uv}^t = \sum_{(v,u)\in EA_t} x_{vu}^t \qquad v = \langle j, o \rangle \in EV, \ t \in T, \ o \in S_t$$
(8.5)

$$\sum_{dv} x_{dv}^{t} = 1 \qquad \qquad t \in T$$

$$(8.6)$$

$$\sum_{(v,d)\in EA_t} x_{vd}^t = 1 \qquad t \in T$$

$$(8.7)$$

$$x_{uv}^t \in \{0, 1\}$$
 $(u, v) \in EA, \ t \in T_{uv}$ (8.8)

Technician temporal constraints.

$$\sum_{(d,v)\in EA_t} y_{dv}^t = \sum_{(d,v)\in EA_t} \tau_{dv} x_{dv}^t \qquad t\in T$$

$$(8.9)$$

$$\sum_{(v,u)\in EA_t} y_{vu}^t = \sum_{(u,v)\in EA_t} y_{uv}^t + \delta_o^t z_v^t + \sum_{(v,u)\in EA_t} \tau_{vu} x_{vu}^t \qquad v = \langle j, o \rangle \in EV, t \in T$$
(8.10)

$$y_{uv}^t \le D^t x_{uv}^t, \qquad t \in T, \ (u,v) \in EA_t \tag{8.11}$$

$$y_{uv}^t \ge 0 \qquad \qquad t \in T, \ (u,v) \in EA_t \tag{8.12}$$

Precedence and synchronization constraints.

$$\sum_{(u,v_1)\in EA_t} y_{uv_1}^t + \delta_{o_1}^t z_{v_1}^t \leq \sum_{(u,v_2)\in EA_r} y_{uv_2}^r + D^t (1 - z_{v_2}^r)$$

$$v_1 = \langle j, o_1 \rangle, v_2 = \langle j, o_2 \rangle \in EV | o_1 < o_2, \ t, r \in T : o_1 \in S_t, o_2 \in S_r$$

$$g_{o_1o_2}^{j\min} - (D^t + g_{o_1o_2}^{j\min})(2 - z_{v_2}^r - z_{v_1}^t) \leq$$

$$\sum_{(u,v_2)\in EA_r} y_{uv_2}^r - (\sum_{(u,v_1)\in EA_t} y_{uv_1}^t + \delta_{o_1}^t z_{v_1}^t) \leq$$

$$g_{o_1o_2}^{j\max} + D^r (2 - z_{v_1}^t - z_{v_2}^r)$$

$$v_1 = \langle j, o_1 \rangle, v_2 = \langle j, o_2 \rangle \in EV | \ t, r \in T : o_1 \in S_t, o_2 \in S_r$$

$$(8.13)$$

Time windows constraints.

$$a_{jo}(\sum_{(u,v)\in EA_t} x_{uv}^t) \le \sum_{(u,v)\in EA_t} y_{uv}^t \le b_{jo}(\sum_{(u,v)\in EA_t} x_{uv}^t), \qquad v = \langle j, o \rangle \in EV(o \neq s_{in}, s_{out}), \ t \in T$$

$$(8.15)$$

Routing of the special device.

$$\sum_{(u,v)\in EA_s} w_{uv}^s = 1 \qquad \qquad v = \langle j, s_{in} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s \neq \emptyset$$
(8.16)

$$\sum_{(v,u)\in EA_s} w_{vu}^s = 1 \qquad \qquad v = \langle j, s_{out} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s \neq \emptyset \qquad (8.17)$$

$$w_{\langle j, s_{in} \rangle, \langle j, o \rangle}^{s} = w_{\langle j, o \rangle, \langle j, s_{out} \rangle}^{s} = 1 \qquad \qquad j \in N, o \in \mathcal{O}_{j} \cap \mathcal{O}_{s} \qquad (8.18)$$

$$\sum w_{s, in}^{s} \leq 1 \qquad \qquad v = \langle j, s_{in} \rangle \in EV, \mathcal{O}_{i} \cap \mathcal{O}_{s} = \emptyset \qquad (8.19)$$

$$\sum_{(u,v)\in EA_s} w_{uv} \le 1 \qquad \qquad v = \langle j, s_{in} \rangle \in Lv, O_j \cap O_s = \emptyset \qquad (8.19)$$

$$\sum_{(v,u)\in EA_s} w_{vu}^s \le 1 \qquad \qquad v = \langle j, s_{out} \rangle \in EV, \mathcal{O}_j \cap \mathcal{O}_s = \emptyset$$
(8.20)

$$w^{s}_{(\langle j, s_{in} \rangle, \langle j, s_{out} \rangle)} \ge \sum_{(u, \langle j, s_{in} \rangle) \in EA_{s}} w^{s}_{(u, \langle j, s_{in} \rangle)} \qquad j \in N, \mathcal{O}_{j} \cap \mathcal{O}_{s} = \emptyset$$

$$\sum w^{s}_{du} = 1$$
(8.21)

$$\sum_{(d,u)\in EA_s} w_{du}^s = 1$$

$$w_{\langle j,s_{in}\rangle,\langle j,o\rangle}^{s} = w_{\langle j,o\rangle,\langle j,s_{out}\rangle}^{s} \qquad j \in N, \ o \in \mathcal{O}_{j} \cap \mathcal{O}_{s} \qquad (8.23)$$

$$\sum_{(u,v)\in EA_{s}} w_{uv}^{s} = \sum_{(v,u)\in EA_{s}} w_{vu}^{s} \qquad v = \langle j,s_{in}\rangle \in EV \text{ or } v = \langle j,s_{out}\rangle \in EV \qquad (8.24)$$

$$\sum_{u,v} w_{ud}^{s} = 1 \qquad (8.25)$$

$$(u,d)\in EA_s$$

$$\sum_{t \in T} x_{uv}^t = w_{uv}^s \qquad (u, v) \in EA_s, u = \langle i, s_{out} \rangle, v = \langle j, s_{in} \rangle, i \neq j$$
(8.26)

$$\sum_{t \in T} x_{(d,\langle j,s_{in}\rangle)}^{t} = w_{(d,\langle j,s_{in}\rangle)}^{s} \qquad (d,\langle j,s_{in}\rangle) \in EA_{s}$$

$$(8.27)$$

$$\sum_{t \in T} x^t_{(\langle j, s_{out} \rangle, d)} = w^s_{(\langle j, s_{out} \rangle, d)} \qquad (\langle j, s_{out} \rangle, d) \in EA_s$$
(8.28)

$$w_{uv}^s \in \{0, 1\}$$
 $(u, v) \in EA_s$ (8.29)

Technician and special device synchronization constraints.

$$\sum_{(u,\langle j,s_{in}\rangle)\in EA_s} y_{u\langle j,s_{in}\rangle}^s = \sum_{t\in T} \sum_{(u,\langle j,s_{in}\rangle)\in EA_s} y_{u\langle j,s_{in}\rangle}^t \qquad \qquad j\in N$$
(8.30)

$$\sum_{(u,\langle j,s_{in}\rangle)\in EA_s} y_{u\langle j,s_{in}\rangle}^s \le \sum_{(u,\langle j,o\rangle)\in EA_t} y_{u\langle j,o\rangle}^t + D^s (1 - z_{\langle j,o\rangle}^t)$$
(8.31)

$$\sum_{(u,\langle j,o\rangle)\in EA_t} y_{u\langle j,o\rangle}^t + \delta_o^t z_{\langle j,o\rangle}^t \le \sum_{r\in T} \sum_{(u,\langle j,s_{out}\rangle)\in EA_r} y_{u\langle j,s_{out}\rangle}^r \xrightarrow{j\in N, t\in T, o\in \mathcal{O}_j\cap\mathcal{O}_s\cap S_t} (8.32)$$
$$j\in N, t\in T, o\in \mathcal{O}_j\cap\mathcal{O}_s\cap S_t$$

$$\sum_{\substack{(u,\langle j,s_{in}\rangle)\in EA_{s} \\ y_{u}^{s} \leq D^{s}w_{uv}^{s}}} y_{u\langle j,s_{in}\rangle}^{s} \leq \sum_{r\in T} \sum_{\substack{(u,\langle j,s_{out}\rangle)\in EA_{r}}} y_{u\langle j,s_{out}\rangle}^{r} \qquad j\in N, \mathcal{O}_{j}\cap\mathcal{O}_{s}=\emptyset \quad (8.33)$$

$$y_{uv}^{s} \leq D^{s}w_{uv}^{s} \qquad (u,v)\in EA_{s} \quad (8.34)$$

$$y_{uv}^{s} \geq 0 \qquad (u,v)\in EA_{s} \quad (8.35)$$

Acknowledgements

We truly thank the Editor and the Reviewers for their constructive critical comments and for having suggested us interesting directions of research that surely improved the first version of this work. Also, we would like to thank V. Pillac, C. Guèret and A. L. Medaglia for providing us with their instances in a truly collaborative atmosphere.

The research of C. Requejo was partially supported by the Fundação para a Ciência e a Tecnologia (Portuguese Foundation for Science and Technology) through project UID/MAT/04106/ 2019 and SFRH/BSAB/142822/2018.

The research of M.G. Scutellà was partially funded by project PRA 2017 of the University of Pisa, "Modelli ed algoritmi innovativi per problemi strutturati e sparsi di grandi dimensioni".

References

- N. Aimonino Ricauda, V. Tibaldi, B. Leff, C. Scarafiotti, R. Marinello, M. Zanocchi, and M. Molaschi. Substitutive "hospital at home" versus inpatient care for elderly patients with exacerbations of chronic obstructive pulmonary disease: A prospective randomized, controlled trial. *Journal of the American Geriatrics Society*, 56(3):493–500, 2008.
- [2] S. Bertels and T. Fahle. A hybrid setup for a hybrid scenario: combining heuristics for the home health care problem. *Computers & Operations Research*, 33(10):2866–2890, 2006.
 Part Special Issue: Constraint Programming.
- [3] D. Bredström and M. Rönnqvist. Combined vehicle routing and scheduling with temporal precedence and synchronization constraints. *European Journal of Operational Research*, 191(1):19–31, 2008.
- [4] P. Cappanera, L. Gouveia, and M. G. Scutellà. The skill vehicle routing problem. In J. Pahl, T. Reiners, and S. Voß, editors, *Network Optimization: 5th International Conference, INOC 2011, Hamburg, Germany, June 13-16, 2011. Proceedings*, volume 6701 of *Lecture Notes in Computer Science*, pages 354–364, Berlin, Heidelberg, 2011. Springer Berlin Heidelberg.
- [5] P. Cappanera, L. Gouveia, and M. G. Scutellà. Models and valid inequalities to asymmetric skill-based routing problems. *EURO Journal on Transportation and Logistics*, 2(1):29–55, 2013.
- [6] P. Cappanera and M. G. Scutellà. Home care optimization: impact of pattern generation policies on scheduling and routing decisions. *Electronic Notes in Discrete Mathematics*, 41:53–60, 2013.

- [7] P. Cappanera and M. G. Scutellà. Joint assignment, scheduling, and routing models to home care optimization: A pattern-based approach. *Transportation Science*, 49(4):830– 852, 2015.
- [8] P. Cappanera, M. G. Scutellà, F. Nervi, and L. Galli. Demand uncertainty in robust home care optimization. *Omega*, 80:95–110, 2018.
- [9] P. Cappanera, M. G. Scutellà, and F. Visintin. Home care services delivery: Equity versus efficiency in optimization models. In A. Matta, J. Li, E. Sahin, E. Lanzarone, and J. Fowler, editors, *Proceedings of the International Conference on Health Care Systems Engineering*, pages 1–13, Cham, 2014. Springer International Publishing.
- [10] E. Cheng and J. L. Rich. A home health care routing and scheduling problem. 1998.
- [11] M. Cissé, S. Yalçındağ, Y. Kergosien, E. Şahin, C. Lenté, and A. Matta. OR problems related to home health care: A review of relevant routing and scheduling problems. *Operations Research for Health Care*, 13–14:1–22, 2017.
- [12] J. Decerle, O. Grunder, A. H. E. Hassani, and O. Barakat. A memetic algorithm for a home health care routing and scheduling problem. *Operations Research for Health Care*, 16:59–71, 2018.
- [13] M. Drexl. Synchronization in vehicle routing a survey of VRPs with multiple synchronization constraints. *Transportation Science*, 46(3):297–316, 2012.
- [14] W. Dullaert, J. Gromicho, J. van Hoorn, G. Post, and D. Vigo. The VeRolog solver challenge 2016–2017. *Journal on Vehicle Routing Algorithms*, 1:69–71, 2018.
- [15] J. F. Ehmke, A. M. Campbell, and T. L. Urban. Ensuring service levels in routing problems with time windows and stochastic travel times. *European Journal of Operational Research*, 240:539–550, 2015.
- [16] L. En-nahli, S. Afifi, H. Allaoui, and I. Nouaouri. Local search analysis for a vehicle routing problem with synchronization and time windows constraints in home health care services. *IFAC-PapersOnLine*, 49(12):1210–1215, 2016.
- [17] P. Eveborn, P. Flisberg, and M. Rönnqvist. Laps care an operational system for staff planning of home care. *European Journal of Operational Research*, 171(3):962–976, 2006. Feature Cluster: Heuristic and Stochastic Methods in Optimization Feature Cluster: New Opportunities for Operations Research.
- [18] FICO Xpress Optimization Suite. http://www.fico.com/en/products/fico-xpressoptimization-suite.
- [19] C. Fikar and P. Hirsch. Home health care routing and scheduling: A review. Computers & Operations Research, 77:86 – 95, 2017.
- [20] S. Frifita and M. Masmoudi. VNS methods for home care routing and scheduling problem with temporal dependencies, and multiple structures and specialties. *International Transactions in Operational Research*, 27:291–313, 2020.

- [21] S. Frifita, M. Masmoudi, and J. Euchi. General variable neighborhood search for home healthcare routing and scheduling problem with time windows and synchronized visits. *Electronic Notes in Discrete Mathematics*, 58:63–70, 2017.
- [22] L. Gouveia. A 2n constraint formulation for the capacitated minimal spanning tree problem. Operations Research, 43(1):130–141, 1995.
- [23] A. Kheiri, A. G. Dragomir, D. Mueller, J. Gromicho, C. Jagtenberg, and J. J. van Hoorn. Tackling a VRP challenge to redistribute scarce equipment within time windows using metaheuristic algorithms. *EURO Journal on Transportation and Logistics*, 8:561–595, 2019.
- [24] G. A. Korsah, A. T. Stentz, M. B. Dias, and I. A. Fanaswala. Optimal vehicle routing and scheduling with precedence constraints and location choice. In *IEEE International Conference on Robotics and Automation (ICRA) 2010 Workshop on Intelligent Autonomous* Systems, May 2010.
- [25] R. Liu, X. Xie, V. Augusto, and C. Rodriguez. Heuristic algorithms for a vehicle routing problem with simultaneous delivery and pickup and time windows in home health care. *European Journal of Operational Research*, 230:475–486, 2013.
- [26] S. Nickel, M. Schröder, and J. Steeg. Mid-term and short-term planning support for home health care services. *European Journal of Operational Research*, 219(3):574–587, 2012. Feature Clusters.
- [27] D. C. Paraskevopoulos, G. Laporte, P. P. Repoussis, and C. D. Tarantilis. Resource constrained routing and scheduling: Review and research prospects. *European Journal of Operational Research*, 263(3):737–754, 2017.
- [28] D. C. Paraskevopoulos, P. P. Repoussis, and C. D. Tarantilis. The resource constrained vehicle routing problem. In *Proceedings of the extended abstracts of the sixth international* workshop on freight transportation and logistics, Odysseus 2015, pages 355–358, 2015.
- [29] V. Pillac, C. Guéret, and A. L. Medaglia. A parallel matheuristic for the technician routing and scheduling problem. *Optimization Letters*, 7(7):1525–1535, 2013.
- [30] M. Rasmussen, T. Justesen, A. Dohn, and J. Larsen. The home care crew scheduling problem: Preference-based visit clustering and temporal dependencies. *European Journal* of Operational Research, 219:598–610, 2012.
- [31] F. Regis-Hernández, G. Carello, and E. Lanzarone. An optimization tool to dimension innovative home health care services with devices and disposable materials. *Flexible Services* and Manufacturing Journal, 2019.
- [32] S. Schwarze and S. Voß. A bicriteria skill vehicle routing problem with time windows and an application to pushback operations at airports. In J. Dethloff, H.-D. Haasis, H. Kopfer, H. Kotzab, and J. SchÅinberger, editors, *Logistics management*, Lecture Notes in Logistics, pages 289–300. Bremen: Springer International Publishing, 2015.

- [33] S. Shahnejat-Bushehri, R. T.-S. Momen, A. Ghasemkhani, and H. Tavakkoli-Moghaddam. Home health care routing and scheduling problem considering temporal dependencies and perishability with simultaneous pickup and delivery. *IFAC PapersOnLine*, 52(13):118–123, 2020.
- [34] M. M. Solomon. Algorithms for the vehicle routing and scheduling problems with time window constraints. Operations Research, 35(2):254–265, 1987.
- [35] S. Yalçındağ, P. Cappanera, M. G. Scutellà, E. Şahin, and A. Matta. Pattern-based decompositions for human resource planning in home health care services. *Computers & Operations Research*, 73:12–26, 2016.